

Received January 6, 2022, accepted January 18, 2022, date of publication January 27, 2022, date of current version February 7, 2022. Digital Object Identifier 10.1109/ACCESS.2022.3147251

Autonomous Convoying: A Survey on Current Research and Development

SAEID NAHAVANDI^{®1}, (Fellow, IEEE), SHADY MOHAMED¹, IBRAHIM HOSSAIN¹, DARIUS NAHAVANDI¹, (Member, IEEE), SYED MOSHFEQ SALAKEN¹, MOHAMMAD ROKONUZZAMAN^{®1}, RACHAEL AYOUB², AND ROBIN SMITH²

¹Institute for Intelligent Systems Research and Innovation (IISRI), Deakin University, Waurn Pond, VIC 3216, Australia ²Future Land Warfare Branch, Australian Army, Canberra, ACT 2600, Australia

Corresponding author: Saeid Nahavandi (saeid.nahavandi@deakin.edu.au)

This work was supported by the Australian Army.

ABSTRACT Convoying or platooning with a fleet of autonomous vehicles, which is denoted as autonomous convoying in this paper, has attracted increasing attention from the research communities, governments, and private sectors in recent years. Autonomous convoying offers immense opportunities due to its potential in enhancing logistical efficiency as well as reducing road incidents/accidents by eliminating human errors due to stress and fatigue. While humans can make complex decisions, involving humans in decision-making processes often causes delays as compared with those of automated machines. Indeed, human errors cause approximately 90% of road accidents and fatalities. Efficient platooning techniques can also reduce fuel consumption and carbon footprints. This paper presents a concise survey on current research and development initiatives in autonomous convoying while critically discussing the underlying techniques and technologies developed in this domain. Implications of autonomous convoying toward different industries are also analyzed and discussed.

INDEX TERMS Autonomy, convoying, sensor fusion, defense, platooning.

I. INTRODUCTION

A convoy refers to a fleet of moving objects (e.g. vehicles or robots) traveling together under the same command. Moving a fleet of vehicles by following one leader decreases the need for human intervention in operations. There is extensive research in truck convoying in recent years [1]. The functionalities of trucks include delivering goods, moving furniture, people, etc. Although there are several options for delivering goods in small locality, such as using cars, motorcycles, or even drones, people mainly rely on large vehicles to deliver food supply and equipment due to cost effectiveness [1], [2]. A convoy of ships is a common technique of transporting goods and military equipment overseas. People often rely on trains in land transportation [3]. However, rail networks are not available in many places, especially where the population density is low. Since trucks move on roads, truck convoying is becoming popular for transporting goods in many areas [4].

The term autonomous driving is synonymously used as self-driving of vehicles or driverless vehicles. In an

The associate editor coordinating the review of this manuscript and approving it for publication was Derek Abbott^(D).

autonomous driving scenario, a vehicle moves autonomously towards its destination. In an autonomous convoving scenario, a group of vehicles moves following a single command and considering the presence of other vehicles in the group [5]. Fig. 1 presents an example of an autonomous convoy of military trucks. Autonomous driving and autonomous convoying are two growing research domains. Fig. 2 and Fig. 3 present the number of published papers related to autonomous driving and autonomous convoying from 2005 to 2019, respectively (statistics gathered from Google Scholar on 17th October 2021). As can be observed, research and development (R&D) in autonomous driving far exceeds that in autonomous convoying, noting that autonomous driving research covers that in autonomous convoying. Since R&D on autonomous convoys requires a larger investment, papers pertaining to autonomous driving are roughly 50 times more than those in autonomous convoving.

In this paper, the term autonomous convoying and autonomous platooning are used synonymously. In a convoy, a group of vehicles travels together, providing mutual support and protection [6]. The word 'platoon' indicates a group of



FIGURE 1. An example of autonomous convoying of military trucks consisting of a leader and two followers (Photograph taken during a project demonstration with the Australian Army).

soldiers receives commands from a leader and acts accordingly. However, autonomous platooning is known in the research community as a group of autonomous vehicles moving by following a leader vehicle [7]. There are research studies that focus on efficient autonomous platooning with better fuel efficiency and better usage of road areas [8], [9]. Several researchers have proposed artificial intelligence (AI)-based solutions for the efficient organisation of vehicles [10]. Small vehicles designed to operate at a high speed are efficiently deployed to minimize air drag. Indeed, air drag costs a significant portion of fuel in large vehicles. Therefore, several research studies investigate the use of machine learning methods for prediction of air drag [11], [12]. It is anticipated that more machine learning-based solutions will be available over time.

Many research initiatives received initial funding for military or exploration purposes. As an example, data compression and aircraft research activities were funded mostly for military and exploration applications in the early stages. The resulting benefits have spread across various fields. In this respect, autonomous convoying, which is an emerging technological development, is currently a key area in military research [13], [14]. Military trucks need to ensure continuous supply for troops and civilians. They are prone to attack from enemies and unexpected situations in contested environments subject to extreme weather or geographic conditions. Robust and reliable autonomous convoying technologies can potentially reduce the number of fatalities, fuel consumption, and deaths due to accident.

Autonomous convoying also offers immense commercial potentials. Several companies have started R&D into realisation of efficient and effective autonomous convoys.



FIGURE 2. The number of publications in autonomous driving in recent years. (Google Scholar, accessed on 17th October 2021).



FIGURE 3. The number of publications in autonomous convoys in recent years. (Google Scholar, accessed on 17th October 2021).

A semi-autonomous fleet of trucks reached the Dutch port of Maasvlakte on 7th April 2016 after traveling across Europe as part of the European Truck Platooning Challenge [4], [15]. In this voyage, trucks remained close to each other, in order to move on roads efficiently and to save time and fuel. The distance between trucks is optimized to minimize air friction. The trucks adjusted the distance between them when any vehicle came into the convoy. Trucks communicated using GPS, Radar, and Wi-Fi, and human drivers were presented in each truck to handle emergencies.

The remaining of this paper is organized as follows. Section II presents autonomous convoying methods and techniques through literature review. Section III presents the capabilities with respect to constructing a successful follower in autonomous convoying. Section IV describes R&D in autonomous convoying in different industries. Section V presents advanced techniques for realizing autonomous convoying. Concluding remarks are given in Section VI.

II. AUTONOMOUS CONVOY TECHNIQUES

Autonomous vehicle convoys or platoons, where each vehicle follows the same path traveled by the preceding vehicle by keeping a safe distance, have become very popular. Autonomous convoying requires the application of several techniques, as described in the later part of this section. It utilizes common path planning methods for keeping a certain formation among all autonomous vehicles. It can potentially solve transportation problems and complete specific tasks more efficiently as compared with the use of single autonomous vehicle. Multi-vehicle convoys are mostly used in military when personnel and resources need to be transported to different locations, and traveling in a convoy provides extra protection for both vehicles and personnel in adverse environments. In addition to military convoys [16], [17], other sectors include commercial vehicle convoys [20].

When a convoy of vehicles travel on the road with ideal weather conditions and clear painted lines, a computer visionbased method can be used to detect the painted road lines and keep vehicles in convoy on the road. However, the images of road lines deteriorate due to weather conditions, obstacles, and missing lines, leading to potential failures in utilizing vision-based control methods [21].

Some autonomous convoys apply a leader-follower approach. In this approach, one vehicle becomes a leader, and other vehicles follow the leader. One can design a follower with lower control complexity in a leader-follower configuration with the help of a leader detection algorithm. Such a follower tracks the leader in an off-road situation and in absence of other vehicles. However, moving under congested traffic conditions and sharp road boundaries requires more consideration. It is possible for the leader to deviate slightly from the expected movement, and uncertainty in the follower system can drive the follower out of the road. When the vehicle convoy is in a city or within certain confined environments in industrial settings, such anerror can lead to unwanted events, e.g. hitting a person or an object, destroying a landmark, etc.

A leader-follower convoy configuration consists of a leader vehicle and one or more follower vehicles. Each follower vehicle obtains the direction signals from its preceding vehicle to stay on the travel route. During convoying, these autonomous vehicles use sensing systems to maintain a safe distance from the front vehicle. Every vehicle in an autonomous convoy knows the path and last stop beforehand as well as the current location along the path in real-time. In this respect, the vehicle to vehicle (V2V) communication method [22]–[24] is useful to share information from the leader vehicle to the convoy system and other follower vehicles. As such, an autonomous convoying system requires a shared travelable path, including a map with path points, poses, speed, and other target tracking information through sensors in each vehicle.

As a full autonomy of the vehicle convoy is yet to be achieved, several research groups are working on different approaches to realize autonomous convoying. Some of the existing methods focus on the global positioning system (GPS) and inertial sensing, which can cause computational latencies due to scene generation,



FIGURE 4. The combined model-based and template-based vehicle tracking architecture [25].

simultaneous localization, as well as real-time object detection and classification. Moreover, methods that use the GPS data of the leader do not necessarily deliver enough accuracy to keep an appropriate convoy traveling path. Inaccuracy in GPS data deteriorates the performance of autonomous convoying. There are some other non-GPS-based approaches where each vehicle tracks its predecessor without resorting to waypoints. However, these methods are not applicable for a large convoy due to accumulative errors with respect to following information [21].

Besides computational latencies, fusion methods are also unable to obtain an acceptable accuracy for long convoys. There are some other drawbacks in using autonomous convoying, including the risk of tracking interruptions, which can be catastrophic for travelling in a convoy. According to [26], an embodiment relating to V2V communication is necessary for allowing safe vehicle convoys by using proper communications between the vehicles [27]. Such embodiment method proposed in [26] enables autonomous vehicles in a convoy to move along a path with a leader vehicle being in communication with at least one of the follower vehicles. The leader can be either in an autonomous mode or in a driver-based mode. The wireless data between vehicles can combined with other vehicle sensor data using methods based on real-time GPS and sensor errors aiming to obtain a precise and safe target tracking analysis of locations for the follower vehicles. Using the proposed method, at least one follower should receive communication pertaining to the target offset position and path data. The target offset position provides a target position relative to the leader vehicle and the path information, including a route history and predicted route of the leader vehicle. The route data can include a global position and a GPS solution of the leader vehicle. Tracking data, including a traveled route of the leader vehicle, can be obtained from sensing equipment of at least one of the follower vehicles [26], [28], [29]. The route data and tracking data are compared, to ascertain accuracy between the route data relative to the tracking data. By observing the tracking data and itinerary data, the target offset location is adjusted. A set of route points is computed to offer a trajectory



FIGURE 5. General framework of autonomous convoy for each vehicle [22].

travel path from an existing location of at least one follower to the adjusted target offset location. Thus, at least one follower can produce the essential control signals to autonomously transition the follower along with the set of trajectory points to the adjusted target offset location.

Petrov [30], Petrov and Boumbarov [31] developed a kinematic model of a two-vehicle convoy and a vehicle following adaptive tracking controller. The follower vehicle combines lateral and longitudinal control to track the trajectory of the leader vehicle. The follower also maintains a desired intervehicle gap. A standard robotic method with homogeneous transformation matrices for modeling the system dynamics is used. The main challenge in developing the controllers for autonomous vehicle tracking is to determine the right trajectory for the follower vehicle. Following the leader vehicle on a curved road segment with a larger or smaller radius and cutting corners is unacceptable.

As the linear and rotational velocities of the leader vehicle and its path curvature radius are unknown constant parameters, the developed adaptive tracking controller using adaptive control law results in asymptotic stabilization of the closed-loop system in error coordinates. The controller receives the relative inter-vehicle positions, orientations as feedback controls as well as approximated velocity of the leader vehicle. With this information, the controller calculates the control velocities of the follower vehicle. With respect to constant velocity maneuvers of the leader vehicle, the two-vehicle convoy can travel concentric arcs of similar radii, keeping a suitable distance between vehicles. This solves the "cutting the corner" issues. For time-varying velocity maneuvers of the leader vehicle, the developed controller in [30], [31] can potentially achieve ultimate boundedness of the closed-loop system in error coordinate. Ollis [21] developed methods for position estimation of an autonomous vehicle convoy. Methods include initializing a convoy state,

choosing the next sensor reading, predicting a convoy state, updating the convoy state, and broadcasting the convoy state to vehicles in a multi-vehicle convoy.

Fries et al. [25] proposed a robust vehicle tracking system for an autonomous convoy in urban and unstructured environments. A monocular camera is adopted as a nonstationary passive vision system. Passive vision sensors are popular as they are inexpensive and have low power consumption. The advantages of model-based tracking systems [32] and template-based tracking methods using various features to accurately estimate a 3D vehicle pose and the associated velocity combined with a fast (re-) initialization approach are evaluated. Fig. 4 presents a combined model-based and template-based vehicle tracking architecture. The model-based method utilizes a particle filter that requires a hypothesis when the system starts and with the presence of a tracking loss. As such, the template-based tracking method needs to re-initialize the particle filter to detect the tracked leader vehicle rapidly. The results of the proposed method show that vehicles that are not properly visible can be tracked in various weather conditions. A visionbased convoy system with a combined pan-tilt-zoom camera mechanism using the monocular camera was developed [16]. The camera keeps the leader in the follower vehicle's field of view while following the path of the leader vehicle. However, a vision-based convoy is not robust against poor visibility and extreme weather conditions.

Zhao *et al.* [22] developed a general framework comprising a real-time leader path following control system, as depicted in Fig. 5. The system consists of pose estimation of the preceding vehicle, leader path queue management, and autonomous controller for convoying. Algorithms for waypoint management, vision & LIDAR-based vehicle tracking, EKF-based data fusion, adaptive inter-distance control, and model-based trajectory with obstacle avoidance are



FIGURE 6. A flowchart for handling size/shape switching [33].

developed. Evaluated on unmanned ground vehicles in different terrains, including off-roads, the results show that the proposed method works efficiently and robustly under environmental disturbances. The system robustness is further enhanced with algorithms for multi-modality fusion-based preceding vehicle trajectory estimation and leader waypoints management. Higher velocities and shorter inter-distances are obtained by applying the developed longitudinal and lateral navigation control method.

Algorithm 1 Reading Horizon Control [33]	
Measure: The current state vector z	-
Calculate: <i>u</i> and <i>z</i> for $[t, t + t_f]$	
Apply: $u = \overline{u}$ for the period $[t, t + \delta]$	
Update: $t \leftarrow t + \delta$	
Return: Start of Size/shape Switching Algorithm	

Algorithm 2 Si	ze Switching	[33]
----------------	--------------	------

Measure: The current state vector zUpdate: d as αd Calculate: u and z for $[t, t + t_f]$ Apply: $u = \overline{u}$ for the period $[t, t + \delta]$ Update: $t \leftarrow t + \delta$ Return: Start of Size/shape Switching Algorithm

]

Jond *et al.* [35] implemented a formation control objective to drive multiple agents for obtaining a pre-defined constraint on their states with a proposed method incorporating

Algorithm 4 Shape Switching (Graph Changed) [33]

Measure: The current state vector z Update: D, W, d Calculate: u and z for $[t, t + t_f]$ Apply: $u = \overline{u}$ for the period $[t, t + \delta]$ Update: $t \leftarrow t + \delta$ Return: Start of Size/shape Switching Algorithm

a quadratic performance index using graph theory [33]. The multi-agent system dynamics are applied as a controllable linear system. The solution of the control law is a nonlinear function of the graph Laplacian matrix and formation of desired distance vectors. Algebraic Riccati equation solution leads to a solution matrix a receding horizon control law in a closed form. A control structure including four algorithms based on closed-form control to handle formation size/shape switching is developed, as shown in Fig. 6. Initially, the convoy follows Algorithm 1. The convoy needs to tune its shape due to obstacles. The convoy also needs to tune its size for any change in the vehicle number. According to the received signals from an external observer or a decisionmaker, the vehicle that is equipped with sensors selects a control mode at each sample time. The control loop stops when the formation control objectives are met. The simulation results of this study indicate efficiency of the proposed solution.

Formation control in an autonomous convoy refers to adjusting the vehicle control inputs to form and maintain a predefined shape and moving configuration. Platooning is a linear formation that mostly considers the longitudinal coordinated control of vehicles [36]. Mohamed-Ahmed *et al.* [37] designed a coupled longitudinal and lateral control method for autonomous convoying using nonlinear predictive control, aiming to track a trajectory. This method enables controlling a convoy using accessible information and following a leader's path while keeping a safe distance between vehicles in the convoy to eliminate any possible collisions. The proposed nonlinear model controls different components. Accelerating and braking wheels control the longitudinal motion of the convoy vehicles. The steering angle is responsible for lateral motions.

The nonlinear control method proposed in [37] provided precision in performance with trajectory tracking of lateral convoy motions. It is also robust when the parameters are not accurately calculated. The control rule ensures a safe distance between vehicles to eliminate any possible collision using longitudinal control while the convoy moves with a similar leader velocity. Accumulation of the convoy tracking error is negligible using the proposed control method. Kato *et al.* [38] developed a new multi-lane platooning method for maximizing safety of cooperative autonomous vehicles on highways. A predefined number of vehicles (up to 5) is engaged in a convoy, including the leader. Using a multi-lane group control method, the level of safety improves as a vehicle collaborates with other vehicles in the same lane

TABLE 1. A summary on autonomous convoying techniques in several published studies.

Paper	Year	Implementation	Ground	Vehicle	Objective	Features / Comment	Affiliation*
[16]	2009	Demonstration	On- & off-road	Robot	Development of a vision-based robotic follower system	One of the best works during the time of its publication. The approach has common limitations of tracking algorithms that rely only on the visual.	Defence R&D, Canada
[17]	2004	Military test	On- & off-road	Robot	Development of both 1) Perceptive follower and 2) Delayed follower.	The perceptive follower stays close to the leader, The Delayed follower follows previously recorded path.	Thales Land & Joint Systems, France
[18]	2003	Simulation	On-road	Truck	Development of both 1) Perceptive follower and 2) Delayed follower.	They have demonstrated several issues of long vehicles in the simulation environment.	University of Illinois, USA
[34]	2003	Demonstration	Highway	Car	Laser based autonomous lateral vehicle following control.	As they are using laser, their algorithm is robust against light fluctuation.	University of California, Berkeley, USA
[20]	2015	Demonstration	On-road	Car	Autonomous convoy driving in absence of light.	Uses internal navigation system (INS), camera, and LiDAR. Capable of estimating the relative 3D position and orientation, the velocity and the steering angle of the leader.	University of the Bundeswehr, Munich, Germany
[21]	2014	Design	On- & off-road	Any	Autonomous convoy with several capabilities: 1) Updating parameter model 2) Processor, memory, communication interface. 3) Set of instructions.	No demonstration or simulation in this work. Only the system design.	Neya Systems, USA
[22]	2017	Demonstration	Terrain	UGV**	A practical real-time leader's path following control system.	A robust work. Many capabilities were demonstrated. System Framework, Waypoints Management, Pose Estimation, Data Fusion, Obstacle Avoidance, Inter-Distance Control, etc.	China North Vehicle Research Institute, China
[25]	2013	Demonstration	On- & off-road	Car	Robust method for vehicle tracking with a monocular camera.	They present a template-based solution using different features to estimate a 3D vehicle pose.	University of the Bundeswehr, Munich, Germany
[26]	2015	Design	Any	Any	Efficient communication between the leader and the follower.	No demonstration or simulation in this work. Only the system design.	GM Global Technology Operations, USA
[30]	2008	Simulation	Any	Any	Modeling of a two-vehicle convoy. Proving concepts with equations and simulations.	Detailed mathematical modeling. Assuming that the leaders' position, linear/angular velocities are known to the follower. There might be uncertainty in real scenarios.	Technical University of Sofia, Sofia, Bulgaria

* Affiliation of the first author.

** Unmanned Ground Vehicles.

as well as neighboring lanes. Marjovi *et al.* [39] developed a technique where a convoy of autonomous vehicles adapts to the path on a road, and different vehicles can join or leave the convoy using a complex messaging system. However, the convoy fails to adapt to changes in terms of road shapes and impacts from other vehicles. Both longitudinal and lateral convoy algorithms are coordinated, since they are necessary for different vehicle maneuvers, including adaption to road shapes, lane changes, as well as maneuvering and overtaking operations [40].

Many researchers apply various multi-agent system-based algorithms in formation control of autonomous convoys. The formation control approaches include vision-based leaderfollower [41], heterogeneous line formation using virtualstructure [42], distributed graph-based control [35], and behavior-based methods [43]-[46]. The leader-follower formation control method for autonomous convoying is one of the most popular methods. However, in case of loss of the leader vehicle, the entire convoy formation fails. To solve this issue, a leaderless formation control mechanism was proposed by Jond et al. [33]. Using an optimal formation control strategy for structuring a leaderless autonomous convoy, each agent has the same role in the group. The control problem of an autonomous convoy is modeled in terms of linear dynamics. The convoy formation control problem is studied under the receding horizon Linear Quadratic (LQ) optimal control framework. The LQ modeling approach to formation control is used because of the analytic tractability of LQ problems. Using the proposed method [33], the formation control objective to drive multiple agents for obtaining a predefined constraint on their states is implemented by applying a quadratic performance index with graph theory [47].

The study in [5] proposed a tracking control method of a convoy of autonomous vehicles to avoid any possible collisions while following a pre-defined path. A coordinate transformation method is used to transform the position errors with respect to each of two consecutive vehicles in a convoy. Transformation is performed from the earth-fixed frame of two consecutive vehicles into their relative position and angle errors to fulfill the predefined transient and steady-state constraints. Using the pre-determined performance method, a nonlinear transformation is used for transforming the constrained relative position and angle errors and achieving an unconstrained kinematic error equation. A new kinematic controller is developed to meet the transient and steady-state performance conditions without causing collisions and controller singularities. Moreover, the Dynamic Surface Control method is applied for simplifying the controller design of the convoy at the dynamic level by applying a first-order filter. An adaptive neural network is employed to keep robustness of the control system against model errors, noise, and disturbances. The Lyapunov theory is exploited to confirm stability of the control system. The overall controller performance for the convoy, including several autonomous vehicles, is tested and validated in a simulation environment, and efficiency of the proposed method is confirmed. A graph-based distributed

control approach in a coordinate system parallel to the path was proposed for controlling heterogeneous vehicles aiming to form multi-lane convoys [35]. Every vehicle in the convoy keeps a local graph with information received from close vehicles while the required distances between vehicles are calculated dynamically. This permits quick adaptation to the variations in vehicle numbers and their locations. Through the implemented distributed mechanism, the vehicles in the convoy can collaboratively change lanes. The formation velocity through this method is fixed, and does not satisfy the shifting necessities for various traffic conditions.

The potential field approach is one of the main methods that create corresponding repulsive and potential fields for obstacles, road maps, and target locations. This is a useful approach to model the driving area, measure risk in the driving environment, and enhance the obstacle avoidance capability. A smooth track without any collisions is planned through the potential field approach while keeping the vehicles close to the initially planned path. Gautam et al. [48] developed a novel technique for a group of multiple robots to establish a chain form from the beginning position to the end position by combining the A*(A-star) algorithm. The technique performs path planning based on a static map and the potential field technique to avoid collisions with obstacles. Huang et al. [49] applied a combination of the potential field technique and model predictive control (MPC), instead of the gradient descent technique, for optimal path planning and vehicle control. However, the potential field-based techniques do not consider speed planning, which is a significant factor in autonomous driving, owing to limitations of the convoy line length, long-distance communication, and incremental time delay in the traditional configuration of vehicle platoons can cause instability. According to Gao et al. [50], distributed graph and potential field offer an effective approach to multilane convoy control. The formation changes its strategy using a distributed graph algorithm to enhance adaptability, obstacle avoidance capability, and stability in various traffic conditions. The potential field approach is applied to build the traffic field model, measure the driving environment risk, and enhance the obstacle avoidance capability to complete motion planning. A double-layer controller for distributed vehicle control is implemented. The higher layer is in charge of path planning and speed planning, and the associated layer formation is a graph model for the road map, with the consideration of obstacles. The lower layer is a tracking control layer that performs lateral and longitudinal control. This layer uses vehicle kinematics to conduct speed control and path tracking for the vehicles. Validated in a simulation environment, the results show efficiency of the proposed method, which possesses better capacity and stability while considering unknown vehicle numbers that can communicate with each other locally.

III. CAPABILITIES OF A SUCCESSFUL FOLLOWER VEHICLE

According to our literature search, a successful follower vehicle requires the following capabilities.

A. FOLLOW THE LEADER VEHICLE

The first requirement of a successful follower vehicle is the ability to track and trail the leader vehicle. There can be a desert of vast plain land without any obstacles and traffic rules. The most rudimentary follower vehicle needs to follow the leader vehicle in situations with no obstacles in the path and no traffic rules.

An example of such a simple follower robot can be toy robots following or tracking a colorful ball [51]. In 1998, Sony Corporation [52] developed a toy dog named AIBO, which can track pink or yellow balls. No hard rules like road boundaries, traffic signals, and fear of major accidents exist in their demonstration.

B. FOLLOW THE LEADER VEHICLE IN INTERSECTIONS

Even a naive autonomous vehicle should be able to follow another vehicle in a straight path with a zero steering angle. A robust autonomous vehicle should not fail to follow the leader vehicle in intersections. The follower vehicle needs to stay within the road while turning in the intersection. Moreover, a long follower vehicle needs a certain maneuver for turning.

When a vehicle convoy transverses along a path in a city, it is expected to encounter traffic lights, which can cause the follower vehicle to lose the leader vehicle. As such, the follower vehicle needs to be able to join the convoy after separation. Several researchers have developed robust follower techniques and applied them in simulation environments [53] or real-world conditions [54].

C. COLLISION AVOIDANCE CAPABILITY

Many researchers have prescribed a separate collision avoidance module for an autonomous system [55], [56]. A follower module can fail to consider details of the road, other vehicles, landmarks, and pedestrians, leading to potential accidents. A dedicated emergency module can help the vehicle to avoid many undesired incidents [54].

D. PEDESTRIAN AVOIDANCE CAPABILITY

An autonomous vehicle needs to detect pedestrians and other moving obstacles. As an example, the autonomous vehicle needs to detect and wait when a pedestrian crosses a road. Another scenario is when a person stands on the side of a road for a long time, the autonomous vehicle needs to detect and avoid the pedestrian [57].

E. CONSIDERATION OF STANDING AND MOVING VEHICLES

An autonomous vehicle needs to travel through different types of roads. One possible scenario is the detection and understanding the condition of a vehicle near the corner of a road. Since there is a possibility for a parked vehicle to appear at the corner of a road, the autonomous vehicle needs to know whether another vehicle is moving or otherwise. If it is a slow-moving vehicle, the autonomous vehicle also needs



FIGURE 7. Negotiation between vehicles at an intersection: An autonomous follower truck needs to detect other vehicles at an intersection and negotiate with them (Photograph taken during a project demonstration with the Australian Army).



FIGURE 8. Driving on a dirt road: An autonomous convoy needs the capability of operating on not just quality tarred roads but also dirt roads without markings (Photograph taken during a project demonstration with the Australian Army).

to detect the movement of that vehicle for planning its own maneuverability [54], [58].

F. ABLE TO NEGOTIATE IN INTERSECTION

The negotiation capability of an autonomous vehicle at an intersection is critical. Based on the position of other vehicles, the autonomous vehicle needs to decide. However, negotiating at the intersection is the task of the leader vehicle. A follower vehicle does not need to negotiate at an intersection unless it is separated from the leader vehicle by traffic signals [59], [60]. Fig. 7 presents a scenario of an intersection, where an autonomous truck needs to negotiate with another vehicle.

G. ABIDING ROAD CONVENTION

An autonomous follower vehicle needs to obey road conventions of a country. Depending on the regulations of a country, vehicles travel either on the left or right lanes. As such, an autonomous vehicle needs to consider vehicles coming in the opposite direction. Road conventions are also required while overtaking other vehicles or allowing overtaking by other vehicles [61], [62].

H. IDENTIFYING OVERTAKING VEHICLES

When a convoy of vehicle transverses along a road, its speed can be slow and a vehicle may want to overtake the convoy. The message pertaining to existence of such a vehicle has to be conveyed from the follower vehicle to the leader vehicle. Moreover, all follower vehicles should not confuse the overtaking vehicle as an obstacle [54], [63]. In addition, heavy autonomous vehicles do not change lanes frequently. So, the convoy of autonomous vehicles needs to consider the overtaking vehicle can appear either from the left or right lane.

I. LENGTH OF A VEHICLE CONVOY

A vehicle convoy can consist of many autonomous trucks. Companies have demonstrated convoys of up to ten trucks. When a vehicle overtakes, another vehicle from the opposite direction can appear [54]. The vehicle convoy can face a change with respect to the road width during the course of traveling. The overtaking vehicle can come to the same lane as that of the vehicle convoy. The relevant follower vehicle should give way for the vehicle to enter the lane. The overtaking vehicle can start another attempt later. As such, the follower vehicle needs to keep communicating with the leader vehicle with respect to maneuverability of the overtaking vehicle. The follower vehicle can adjust and maintain an appropriate distance when the overtaking attempt commences again.

J. SENSING THE ROAD PROPERLY

An autonomous vehicle needs to sensorise a road properly. When a road has multiple lanes, the autonomous vehicle needs to detect the available lanes. The algorithm and sensor systems have to be robust enough to detect the road scene under varying weather conditions, presence of other vehicles and pedestrians, or occlusions that can degrade detectability of the road conditions [64].

K. DRIVING ON DIRT ROADS

A vehicle convoy can be useful for sending goods to rural and remote locations, which may not have painted roads. In many contested conditions, autonomous trucks need to travel on dirt roads or mountainous tracks [65]. The implementation of a robust autonomous convoys in such conditions is beneficial. Fig. 8 depicts an autonomous convoy travelling on a dirt road.

L. NIGHT OPERATIONS

A robust autonomous vehicle needs the path following capability, even in the absence of light. Night operations can be under lit or unlit conditions [54]. As such, night operations of an autonomous vehicle need the assistance of information from multiple sensors. Choi *et al.* collected GPS, LiDAR, stereo, thermal image, and RGB stereo images to produce a combined data set for night operations of autonomous vehicles [66].

M. DYNAMIC REROUTING

Typically, an autonomous vehicle performs a rough path planning before starting its journey. Due to obstacles and potential hazards, the vehicle needs to dynamically change



FIGURE 9. Proof of autonomy by lifting hands by a driver (who takes over vehicle control only during emergencies) while the military truck moves on a dirt road (Photograph taken during a project demonstration with the Australian Army).

its routes when the initially planned path is unavailable [67]. The capability of dynamic routing, which is the process of selecting an optimal path given the current traffic conditions, is therefore crucial in autonomous convoying.

N. CONSIDERATION OF THE VEHICLE LENGTH

Research studies on autonomous cars attempt to narrow the vehicle length, since the driving seat is not required. However, in autonomous convoying with trucks, the vehicle length is a key concern [68] because maneuverability, speed limits on intersections, and overtaking considerations are different.

O. BOTH FORWARD AND BACKWARD MOVEMENTS

A follower vehicle may need to perform a backward movement in unforeseen situations. Any blockade on the road can cause the entire autonomous convoy to move in the backward direction until the leader vehicle reaches a previous intersection. Many goods loading and unloading spots are located at the end of a road. Backward driving is different for long vehicles due to the jack-knife effect [69]. This backward movement capability needs to avoid collision with objects, pedestrians, and other vehicles. Therefore, a follower vehicle needs the ability to move in a backward direction on straight roads, bent roads, intersections, and other conditions robustly, efficiently, and safely.

P. ROBUST COMMUNICATION

A robust and secure short-range communication capability among autonomous vehicles is important. Communication should not be affected by other vehicles or any malicious intrusions. Ucar *et al.* [70] proposed a visible-light hybrid communication-based platoon using the front light of the follower vehicle and the tail light of the leader vehicle. The IEEE 802.11p radio frequency communication technology is also adopted, in order to produce a secure communication system.

Q. PROOF OF AUTONOMY

As autonomous convoying is in its infancy, companies keep a driver at the driving seat of the follower vehicles. This scenario raises the question whether the follower vehicles can move autonomously. As shown in Fig. 9, drivers raise their hands from the steering wheel to prove that the follower vehicle moves autonomously. The driver's duty is to take over control of the vehicle during emergencies [54].

IV. R&D ON AUTONOMOUS TRUCK CONVOYING IN DIFFERENT INDUSTRIES

According to our literature search, the following industries are significantly contributing towards R&D on autonomy and autonomous convoying of trucks.

A. AUTONOMOUS MOBILITY APPLIQUÉ SYSTEM (AMAS) BY LOCKHEED MARTIN

The AMAS Leader-Follower mode links a large fleet of vehicles together as a cohesive convoy. As a result, the follower vehicles can operate without a person in the driver's seat [54]. Lockheed Martin conducted a military user assessment in Carolina in the summer of 2014, and demonstrated driverassist capabilities in December 2015. The recent solution can perform leader-follower convoying with consideration of moving obstacles and presence of small cars in the middle. These capabilities have been evaluated in both urban and rural environments and low-light conditions. As indicated in the AMAS report, current challenges include barriers and obstacles, rights of way, dynamic re-routing, 4-way intersections, pedestrian traffic, emergency braking, and negotiation in a traffic circle.

Although Lockheed Martin has achieved great progress in convoy autonomy, there exists some confusion in their demonstrations. Negotiation with pedestrians and other vehicles in intersections is still challenging for humans. Humans struggle to decide which vehicle should go first. However, autonomous vehicles can solve this issue through communications. When a human dummy approaches, the truck can move on the other side of the road, or the truck can wait until the dummy crosses the road. Lockheed Martin has not disclosed any clear margin between these two decisions.

B. LOCOMATION

A Missouri-based truck company, namely Locomation, has started a multi-year program on human-guided autonomous



FIGURE 10. A follower truck follows the leader truck in leader-follower autonomy. The requirement of a driver is reduced by almost 50% and the fuel cost is reduced by 8%. The reason of reduction on fuel consumption is a lower air drag in the follower truck. This picture is drawn according to the description of the Locomation convoy in [71].

convoying. According to the American Trucking Association, in 2018, trucks transported 11.5billion tons of freight and made nearly 800 billion USD in revenue. The demand for autonomous convoying technologies has increased due to COVID-19. Locomation announced that they are working with Wilson Logistics to transport cargo using autonomous trucks, e.g., covering more than 400 miles between Oregon and Idaho in each journey [71]. The company has investigated the use of autonomous follower trucks to deliver goods with less human involvement. This solution also minimizes air drag and optimizes fuel efficiency [72]. Fig. 10 presents an air-drag minimization scenario.

Vehicles share data on steering angle, acceleration, speed, and the degree of brake and throttle applied. Locomation trucks use an intricate system of radar, lidar, and cameras to observe the surroundings. To avoid blind spots, the trucks are well-equipped with a range of sensors. A place behind the driver's seat is suggested for the driver of a follower truck to rest and sleep after activating autonomous operations. Currently, a driver sits on the driver's seat when automation is enabled. In the future, drivers can leave the seat to take a rest. According to Avi Geller, the CEO of Maven Machines, a comprehensive management system for the fleet is required, as the vehicles are becoming more sophisticated. In the next five years, the driver assistance features may eventually become standard in industries. Locomation has received about 5.5 million USD of funding from different sources. A plan to add the third truck is in place. If they can add the third truck, the driver requirement can be reduced to about 33%. Locomation currently evaluates their autonomous convoying system at the Transport Research Center in Perry.

Many economic issues related to autonomous truck convoying need to be considered. As an example, issues such as whether the driver of the follower truck should be paid during his/her rest period, equal payment for both active and inactive drivers, etc. need to be addressed.

The steps of achieving semi-autonomy are as follows:

- Two drivers sit in different trucks. They start the journey, cross the surrounding urban/busy area with manual driving.
- Then, one truck takes the lead and the other follows.

• Truck drivers can change rules. Trucks can interchange their position, and the leader truck driver can take a rest while the follower truck driver takes over the driving duty.

C. EUROPEAN TRUCK MANUFACTURERS

Volvo is currently evaluating autonomous garbage collection trucks in urban environments [73]. In addition, multiple manufacturers, which include Iveco, MAN, Scania, Volvo Group, DAF, and Daimler [74], have established plans to demonstrate multi-brand autonomous truck convoying on European roads from 2018 to 2022.

D. GOOGLE

A Google spinoff, i.e., Waymo, made use of autonomous trucks to deliver goods to the data centres of Google in Atlanta in 2018 [75]. Currently, they are planning to extend their services to Texas and Mexico, due to commercial potentials. The company conducts test and evaluation in Huston, Dallas, and El Paso. In Mexico, the company perform tests in the southern part of the country. In June 2019, Waymo engaged 13 robotic experts, including Boris Sofman, former CEO and co-founder of Anki, to provide expertise to the company.

E. NATIONAL ROBOTICS ENGINEERING CENTER (NREC) AT CARNEGIE MELLON UNIVERSITY

NREC provides innovative robotic solutions [76]. Recently the center is working on a cargo unmanned ground vehicle. Several capabilities on unmanned ground vehicles have been established, as follows:

- 1) Vehicles must have instantaneously switching capability among normal, remote, and autonomous driving.
- 2) Vehicles must be aware of other moving vehicles, vehicles in a convoy, and moving pedestrians.
- 3) Vehicles should drive safely in varying conditions including the presence of smoke, dust, snow, and rain.
- 4) Vehicles must navigate robustly with respect to any visual degradation of the path caused by a previous vehicle or any previous event.
- 5) Vehicles must be capable of operating in different environments, such as open country, mountains, villages, and cities.

An autonomous vehicle requires a good perception capability pertaining to visual objects and their speed. Several planning mechanisms are also necessary, namely:

- 1) Local planning: generating a short-term trajectory based on recent information to help the autonomous vehicle in lane-following and obstacle avoidance;
- Global planning: computing the shortest path based on the available global map;
- 3) Speed planning: setting the autonomous vehicle speed based on the road condition and visibility.
- 4) Traffic planning: interacting with traffic and other moving obstacles near the road.



FIGURE 11. Road area, vehicle, and obstacles are merged into a single map (drawn according to the information from NREC [76]). White and green regions represent previously and newly re-explore areas in the map, respectively. An existing map for initial path planning, is required and recent changes of information in the map are helpful in obstacle avoidance and rerouting.

5) Learning: applying machine learning to improve performance.

A local map containing information on previously established areas and newly explored areas is crucial. Fig. 11 depicts a visualization on sensor and map fusion. In Fig 11, the green areas of the map is newly explored regions through sensors, which are merged with previously recorded areas in the map.

F. BAE SYSTEMS

According to a recent report, key vendors in military autonomy are BAE Systems, Elbit Systems, Lockheed Martin, Northrop Grumman, Polaris Industries, and RUAG Group [77]. BAE System is one of the top companies in the business of land artilleries. The company has demonstrated autonomous battlefield unmanned ground vehicles to the Australian Army. However, documents on autonomous convoying capabilities developed by BAE Systems are unavailable publicly.

G. MILITARY RESEARCH IN AUTONOMOUS CONVOYING

Many research areas started with initial sponsorships for defense and exploration purposes. As an example, Shannon's

theory and related techniques were heavily sponsored by the space exploration industry in the 1960s. The benefits of these research studies have now been extended to many current compression algorithms [78]. Similarly, while R&D on airplane was initiated by military forces in its early stage [79], air travelling now brings benefits to everyone.

In 2018, the US Army planned to deploy seventy self-driving trucks for supply chain activities, with human supervision in 2020 and full autonomy by 2022 [80]. A human driver is engaged to operate the leader vehicle, while unmanned follower trucks should not lose the leader vehicle in the absence of a clear vision due to rain, snowfall, dust, etc. While humans are very good at driving in such situations, the use of computer vision for autonomous driving is still in its infancy which can fail to act properly in noisy environments. In contested environments, the follower trucks need to consider shell holes, rocks, trees, rubble etc., as well as to recognise animals, pedestrians, and other vehicles.

The US Army is confident in the development of computer vision. The technology can remove manpower from becoming the front-line troops, therefore maximising safety of soldiers. The US Army planned to investigate ten Oshkosh M1075 PLS (Palletized Loading System) trucks and convert 60 more vehicles to self-driving vehicles by 2020 [80].

Note that the leader-follower convoy concept of the US Armay is not in a full autonomy setting. Full autonomy in a convoy of vehicles is expected in 2022. The experiments from 2020 to 2021 are based on the leader-follower combination, where humans are present in the driver seat to handle emergencies. While identifying an optimal path or a moving direction is not difficult, travelling with robust obstacle avoid-ance capabilities is a great challenge. Autonomous vehicles are not free from accidents, but their performance can be better than humans in the future.

Autonomous vehicles in cities rely on the use of GPS for navigation. Due to hostile terrains or network jammers that can block the reception of GPS signals [80], military research studies utilise an internal navigation system (INS). The INS consists of an accelerometer, gyroscope, and computer algorithms. These algorithms face higher uncertainty [81] when traveling a long distance. Therefore, military robots need to calibrate their estimations from landmarks. Humans also do similar calibrations when travelling without GPS signals. By observing a landmark, humans become certain with respect to distance travelled thus far and the remaining journey. The crucial part of off-road autonomous vehicle navigation is the combined outputs from radar, camera, and lidar. GPS signals serve as a support means, whenever available. The computer algorithms in a leader vehicle are used to detect landmarks and compute the locations and dimensions of landmarks. Information on landmarks is transmitted to follower vehicles. A good INS system can reduce the required number of landmarks. The INS system in US Army vehicles is very accurate, which has only ten centimeters of error after traveling 100 meters. As such, only few calibrations are required, and the number of required landmarks is also low [82]. The combination of INS and landmarks allows a follower vehicle to follow the path taken by the leader vehicle, even after one month. This raises a question on the requirement of a human driver. According to the US Army, a human driver is still needed to handle situations with barriers and bushes that can cause safety issues.

V. ADVANCED AUTONOMOUS CONVOYING METHODS

While humans are good at controlling vehicles, approximately 90% of road casualties are due to human errors. Moreover, recent advancement in technology has brought higher capabilities that surpass human capabilities. Machines can sense a large number of information, and take collaborative decisions within seconds [83]–[85]. Therefore, researchers introduce advanced methods by combining a range of sensors and computational intelligence to develop new autonomous driving technologies.

A. MULTI-AGENT COLLABORATION

Multi-agent-based solutions are important in tackling problems that need collaborative effort. In Fig. 12, two players (denoted by red squares) collaborate to capture (i.e., occupy) the blue square that can change its location dynamically. Grey regions denote walls. Squares can move only in black regions. Capturing the blue square by one player (one red square) is difficult. An efficient collaborative strategy by two players can reduce the number of movements required to occupy the blue square.

Recently a collaborative Artificial Intelligence (AI) model was applied to bring the grandmaster level performance in the game StarCraft II [86]. StarCraft II is an online strategy game. Players need to build a troop of soldiers, construct buildings, and fight with opponents. In the game, the AI agent learns by imitating human players. When a human defeats the AI agent, the strategy of the human is learned [87]. Multi-agent collaboration is based on the consideration of uncertainties associated with the target. These uncertainties vary from situation to situation [88]. It is possible to derive mathematical formulations for small games, as shown in Fig. 12. Training of neural networks by imitating humans is effective for complex games. However, trial and error are not possible during real wars. Multi-agent collaboration is possible in a small part of a war. We anticipate large-scale multi-agent collaborations [89] in real wars when a significant percentage of troops can operate autonomously.

As research in autonomous convoying is in its early stage, multi-agent collaboration is yet to be investigated by companies. Currently, a vehicle needs a human drive to start and end a journey. In the future, it is expected for a vehicle to reach its loading/unloading location autonomously. The vehicle chooses its position in a convoy autonomously with multi-agent collaboration strategy for successful loading/ unloading activities.

There are several futuristic academic research on autonomous convoying based on multi-agent collaboration. A recent paper proposed rules for automotive platoons [91].



FIGURE 12. An example of multi-agent collaboration (redrawn from an image in Google DeepMind [90]). The objective is for two players (red squares) to collaborate and capture the blue square that can change its location dynamically.

A vehicle can join a convoy either at the end or in the middle position with different control strategies. The joining procedure is as follows:

- 1) A non-member vehicle sends a joining request to the leader vehicle.
- 2) If a vehicle joins a convoy at the rear position, the leader vehicle sends back an agreement, provided that the maximum convoy length is not reached.
- 3) If a vehicle joins a convoy in the middle position, e.g. in front of vehicle Y, the leader vehicle sends an "increase space" command to vehicle Y, and the new vehicle receives an agreement
- Changing lanes to join a convoy is a manual procedure performed by a human driver. Upon receipt of an agreement, the joining vehicle changes its lane.
- 5) Once the vehicle is in the correct lane, its automatic speed controller is enabled to approach the preceding vehicle.
- 6) When the vehicle is close enough to the preceding vehicle, its automatic steering controller is enabled, and an acknowledgment to the leader vehicle is sent; and, finally,
- 7) The leader vehicle sends a "decrease space" command to vehicle Y, and when the leader vehicle is informed that spacing has returned to normal, it replies to the acknowledgment message.

The leaving procedure is:

- 1) A convoy member sends a leaving request to the leader vehicle and waits for an authorisation.
- 2) Upon receipt of a 'leave' authorization, the vehicle increases its space from the preceding vehicle.
- 3) When the maximum spacing is achieved, the vehicle switches both its speed and steering controller to 'manual' and changes its lane; and, finally

4) The vehicle sends an acknowledgment to the leader vehicle.

Several research groups also proposed simulators for automated platooning [92], [93].

B. ROAD SEGMENTATION

A road segmentation model takes an image of a road and provides a binary classification output for every pixel. In the training data, the image pixels are labeled as road and non-road. Multilabel semantic segmentation models are employed to segment various objects in the image. Through multilabel semantic segmentation, an estimation of both the drivable surface and objects on that surface can be obtained. The drivable surface includes all pixels with respect to the road, parking spots, lane markings, crosswalks, and even rail tracks. Objects on that surface can be other vehicles, pedestrians, animals, etc. With the information on the drivable surface, an autonomous driving system constructs an occupancy grid. The occupancy grid is constructed by projecting several points of the LiDAR, or RADAR point cloud on the visual image. The random sample consensus (RANSAC) algorithm is a popular method to robustly fit a drivable surface plane model even with erroneous semantic segmentation [95]. The occupancy grid construction algorithms work well on flat road surfaces. The occupancy grid helps an autonomous vehicle to move with collision avoidance capability.

Researchers have also developed road segmentation algorithms based on different types of input data. The most common approach is road segmentation on RGB images [96], [97]. Recently, methods for segmenting roads with LiDAR and RADAR data have also been developed [98]–[101]. There are several road segmentation data sets that exist in multiple forms. Since many new road segmentation methods yield different results, it is difficult to predict which combination works the best in the future. In a KITTI challenge [102], the best results use both image and depth information. Several studies indicate the presence of depth information increases segmentation efficiency [103]–[105]. Fig. 13 presents an example of a road segmentation method where both RGB image and depth information are considered, in accordance with the study in [94].

C. FUSION

Data fusion techniques can be broadly classified into three categories [106]: (i) state estimation, (ii) data association, and (iii) decision fusion. Data, sensors, and decisions fusions are performed for visualization, reliability, and further processing of information.

1) ALGORITHM AND DATA FUSION

In real-world applications, no sensor information is errorfree. Moreover, detection or classification models are subject to errors. Assume that the first detection system has a success probability of $P(S_1)$. As such, the probability of failure in the *i*th detection system becomes:

$$P(F_i) = 1 - P(S_i).$$
 (1)



FIGURE 13. Based on [94], a robust road detection method that incorporate the convolutional neural networks (CNNs) can be developed to combine camera and LiDAR images [94] (Photograph taken during a project demonstration).

Assume that two sensors or algorithms are fused together, and they are independent of each other. Their combined probability of failure is:

$$P(F_1 \cap F_2) = P(F_1) \times P(F_2).$$
 (2)

The probability of success from both systems is also in a multiplicative form:

$$P(S_1 \cap S_2) = P(S_1) \times P(S_2).$$
 (3)

However, when the system is used for ensuring safety or detecting obstacles, the outcome from one algorithm or system is considered as the correct detection. The detection success probability from any two sensors or algorithms can be written as follows:

$$P(S_1 \cup S_2) = 1 - P(F_1 \cap F_2).$$
(4)

Their combination does not bring success when both of them fail. Combining with Eqn. (2), Eqn. (4) can be written as:

$$P(S_1 \cup S_2) = 1 - P(F_1) \times P(F_2).$$
(5)

Therefore, the equation for getting detection success from the fusion of N systems becomes:

$$P(S_1 \cup S_2 \dots \cup S_N) = 1 - \prod_{i=1}^N P(F_i).$$
 (6)

As $P(F_i) < 1$, multiplying any number with $P(F_i)$ decreases the value, i.e., $P(F_1) \times P(F_2)$ is smaller than both $P(F_1)$, and $P(F_2)$. Therefore, $P(S_1 \cup S_2)$ is greater than both $P(S_1)$, and $P(S_2)$. We can conclude that the probability of success increases with fusion of sensors or algorithms. Moreover, one sensor or computation utility may not work at a certain time. The presence of multiple sensors or algorithms can provide information that allows a greater degree of autonomy in such situations.

Besides detections, algorithms provide prediction of numeric quantities, e.g. steering angle, desired vehicle speed, etc. The mean or median value of numeric quantities can be utilised for further processing. While one prediction system can easily be subject to incorrect outcome, the probability of all prediction systems being incorrect is lower [64]. Moreover, domain experts know the input range of different sensors. They can develop algorithms to select an optimal sensor-algorithm combination under different situations.

2) SENSOR FUSION

Although it is possible to achieve semi-autonomy with single image sensor, such a system may fail in the absence or fluctuation of light condition [107]. Uncertainties exist in information solicited from the environment, sensors, algorithms, and controllers [108], [109]. Therefore, a robust autonomous vehicle is equipped with a range of sensors to minimise uncertainties associated with perception.

The concept of sensor fusion is developed for optimal information processing in multisensory environments [110]. A paradigm of sensor fusion is human perception in food tasting scenarios. Humans perceive food through our eyes, nose, and tongue. Information from all sensors helps a human to identify the underlying ingredients. There are several motivations for sensor fusion, e.g. to accommodate partial or total failure of sensors as well as limited coverage and precision that cause uncertainty. Sensor fusion enables a system to act correctly, in the event of noisy information from certain sensors.

Sensor fusion allows the creation of an internal map that contains various detected objects with different confidence levels. Different sensors have different resolutions and ranges, and they cover different regions. As such, sensor fusion brings benefits pertaining to accurate position, orientation, and situational awareness [111]. There are three general types of sensor fusion: competitive, complementary, and cooperative. In competitive sensor fusion, each sensor derives an independent measurement. A decision is taken based on individual outputs. Competitive sensor fusion brings higher accuracy. In complementary sensor fusion, sensors are combined to provide a more complete definition [112], e.g. images from two cameras can be combined to have a better area coverage. In cooperative sensor fusion, multiple sensors derive combined information that cannot be achieved from using just single or fewer sensors, e.g. understanding 3D structures from multiple image sensors [113]–[115].

Sensor fusion is becoming popular in autonomous driving [116]. An autonomous vehicle uses several sensors, which include camera, radar, LiDAR, odometer, global positioning system (GPS), infrared (IR) devices. Information from a camera is usually useful in the presence of light. However, a change in lighting condition or any painting on the road can cause errors when the detection algorithm relies on images only. Radar and LiDAR are useful in detecting objects under dim conditions. However, it is difficult to classify objects based on radar and LiDAR signals. An internal map and odometer information can potentially help recognise nearby landmarks, even in the absence of light. The combined information from sensors reduces uncertainties of the perceived environment and improves the confidence of autonomous driving [117]–[120].

Sensor fusion can be used to build an occupancy map by combining signals from different sensors, e.g. in indoor navigation [121]. The first stage of building an occupancy map is to transform and combine data. A 2D image from a camera is different from radar or LiDAR data [122]. Therefore, transformation from the image coordinate to robot coordinate through mapping is required [122]. The second stage is combining the information to enhance robustness. Information from various sources (e.g. image sensor, radar, sonar, etc.) is combined using competitive, complementary, or cooperative sensor fusion methods. Algorithms are applied to the combined data for safe navigation. Indeed, sensor fusion makes the information more reliable and reduces the chance of a collision or failure [123]. Obstacles in the occupancy map is generally made larger, in order to prevent potential collisions by autonomous agents [122].

Sensor fusion is a growing field in military applications. The market of military sensor fusion is predicted to grow to 756 million by 2029 [124]. Indeed, sensor fusion has been deployed in various military equipment in several countries, which is useful in target detection, combat vehicles, enemy identification, weapon systems, etc. Combining GPS data with information from an odometer, magnetic compass, gyroscope, and map results in a more accurate occupancy map [125]. Both the position and orientation information of an equipment is required when navigating under dim conditions. Identifying the positions of other moveable objects is important in autonomous convoying, e.g. other vehicles, peoples, enemies, or animals in specific locations. Currently, leading companies that develop military sensor fusion technologies are General Dynamics, Esterline Technologies, Safran Group, BAE Systems, Lockheed Martin Corporation, Honeywell International, and Kongsberg Gruppen [126].

D. FUEL AND SPACE EFFICIENCY

Fuel and space efficiency can potentially be achieved through a carful platooning strategy. Recently, Locomation demonstrated that minimization of air drag can lead to higher fuel efficiency. Researchers have also proposed several advanced techniques for efficient air drag control [127]–[129] and road space utilisation [130]–[133]. Both fuel and space efficiency can also be achieved through efficient traffic signaling [134]–[136]. It is envisaged that extensive research from industries to improve fuel and space efficiency will emerge in the near future.

E. MODULAR VS END-TO-END APPROACHES

There are two popular approaches to autonomous driving [137], [138]: modular and end-to-end methods. Fig. 14 presents an end-to-end approach. Autonomous vehicles use multiple sensors for ensuring a successful operation, e.g. infrared (IR) sensor, LiDAR, RADAR, ultrasonic, odometer, camera, etc. Besides sensors, autonomous vehicles have dedicated short range communication (DSRC), internal navigation system (INS), map, and global positioning system (GPS) to improve sensing of the environment.

The modular approach is widely applied in industries [139], with the advantage of ease of implementation [140]. An accident can occur due to a fault in a sensor or an algorithm. The algorithm may work with one or more faulty sensors. Several algorithms are applied for autonomous driving in the modular approach, e.g. algorithms for traffic signal detection [141], pedestrian detection [142], road segmentation [143], steering angle prediction [144], speed prediction [145], and path planning [146]. When the sensor readings and recent predictions are recorded, it becomes easy to investigate and improve the performance of different modules.

The end-to-end approach can potentially bring good performance with sufficient samples and proper training methodology [147]. Many researchers anticipate that the end-to-end approach will become better with the advancement in AI [148].

F. FUTURE OPPORTUNITIES FROM AI

AI is a fast-growing research domain, and an immense improvement in AI methodologies can be observed in the last decade [149]–[151]. AI provide smart solutions to many existing problems, which has brought new research dimensions [152]–[154]. With the help of AI, a novice user can develop prediction models. The advent of transfer learning models yield good performance with reduced computation loads and smaller data sets [155]. Recent advancements in DeepFake have caused difficulty in distinguishing real and fake video clips by humans. Successful autonomous driving requires many calibrations [156], [157], e.g. slight differences between the odometer reading and the actual distance traveled by a vehicle normally exist [158], [159]. In this case, fusing sensor information or adjusting sensor readings requires calibrations [160]. Neural networks have been used as a useful



FIGURE 14. Sensor fusion with an artificial intelligence (AI) model for end-to-end autonomous driving, where the AI model receives information from different sensors and predicts the relevant outputs.

method for such calibrations [161], which benefit R&D in autonomous convoying.

VI. CONCLUSION

Autonomous convoying is a rapidly growing research field to date. Convoying with a fleet of autonomous vehicles requires the consideration of many aspects. This survey has provided important information on autonomous convoys to facilitate future R&D activities. We have analyzed current methods and techniques associated with autonomous convoying, covering tracking and control mechanisms, characteristics of a good follower vehicle and related real-world scenarios. This literature also has presented current R&D initiatives of different industries and future possibilities. As the technology of autonomous convoying is in still its early stage, many research activities are funded by the military at this moment. Successful implementation of a fully autonomous convoy can potentially enable transporting goods to remote places without human involvement. Indeed, it is envisaged that logistic and supply chain activities to help people in rural and remote locations with greater efficiency without endangering life become possible through realization of autonomous convoying in the future.

REFERENCES

- Robots Take to the Streets. Accessed: Sep. 2, 2020. [Online]. Available: https://www.fleetpoint.org/autonomous-vehicles/robot/robots-take-to-the%-streets/
- [2] Self-Driving Warship, 'Smart Trucks' to Revolutionise Military Operations, Road Transport. Accessed: Sep. 2, 2020. [Online]. Available: https://www.abc.net.au/news/2016-04-08/are-self-drivingwarships-and-sm%art-trucks-the-future/7310336
- [3] C. McLennan, Big Military Convoy Pulls Into Town. Video and Pictures. Accessed: Sep. 2, 2020. [Online]. Available: https://www.katherinetimes. com.au/story/5919474/big-military-convoy-pul%ls-into-town-videoand-pictures/
- [4] Self-Driving Truck Convoy Completes its First Major Journey Across Europe. Accessed: Sep. 2, 2020. [Online]. Available: https://www. theverge.com/2016/4/7/11383392/self-driving-truck-platoonin%geurope

- [5] K. Shojaei and M. R. Yousefi, "Tracking control of a convoy of autonomous robotic cars with a prescribed performance," *Trans. Inst. Meas. Control*, vol. 41, no. 13, pp. 3725–3741, Sep. 2019.
- [6] D. Iozsa, C. Stan, and L. Ilea, "Study on the influence of the convoy rolling over aerodynamic resistance," *IOP Conf. Ser., Mater. Sci. Eng.*, vol. 252, no. 1, 2017, Art. no. 012035.
- [7] A. Ghosal, S. U. Sagong, S. Halder, K. Sahabandu, M. Conti, R. Poovendran, and L. Bushnell, "Truck platoon security: State-of-the-art and road ahead," *Comput. Netw.*, vol. 185, Feb. 2021, Art. no. 107658.
- [8] M. P. Lammert, B. McAuliffe, P. Smith, A. Raeesi, M. Hoffman, and D. Bevly, "Impact of lateral alignment on the energy savings of a truck platoon," Nat. Renew. Energy Lab. (NREL), Golden, CO, USA, Tech. Rep. NREL/CP-5400-78216, 2020.
- [9] B. McAuliffe, A. Raeesi, M. Lammert, P. Smith, M. Hoffman, and D. Bevly, "Impact of mixed traffic on the energy savings of a truck platoon," *SAE Int. J. Adv. Current Practices Mobility*, vol. 2, pp. 1472–1496, Apr. 2020.
- [10] G. A. P. de Morais, L. B. Marcos, J. N. A. D. Bueno, N. F. de Resende, M. H. Terra, and V. Grassi, Jr., "Vision-based robust control framework based on deep reinforcement learning applied to autonomous ground vehicles," *Control Eng. Pract.*, vol. 104, Nov. 2020, Art. no. 104630.
- [11] F. Jaffar, T. Farid, M. Sajid, Y. Ayaz, and M. J. Khan, "Prediction of drag force on vehicles in a platoon configuration using machine learning," *IEEE Access*, vol. 8, pp. 201823–201834, 2020.
- [12] M. Usama, A. Arif, F. Haris, S. Khan, S. K. Afaq, and S. Rashid, "A datadriven interactive system for aerodynamic and user-centred generative vehicle design," in *Proc. Int. Conf. Artif. Intell. (ICAI)*, Apr. 2021, pp. 119–127.
- [13] S. Grigorescu, B. Trasnea, T. Cocias, and G. Macesanu, "A survey of deep learning techniques for autonomous driving," *J. Field Robot.*, vol. 37, no. 3, pp. 362–386, 2020.
- [14] B. R. Kiran, I. Sobh, V. Talpaert, P. Mannion, A. A. Al Sallab, S. Yogamani, and P. Pérez, "Deep reinforcement learning for autonomous driving: A survey," *IEEE Trans. Intell. Transp. Syst.*, early access, Feb. 9, 2021, doi: 10.1109/TITS.2021.3054625.
- [15] L. Aarts and G. Feddes, "European truck platooning challenge," in *Proc. HVTT Int. Symp. Heavy Vehicle Transp. Technol.*, Rotorua, New Zealand, 2016, pp. 15–18.
- [16] J. L. Giesbrecht, H. K. Goi, T. D. Barfoot, and B. A. Francis, "A visionbased robotic follower vehicle," *Proc. SPIE*, vol. 7332, Aug. 2009, Art. no. 733210.
- [17] L. Vasseur, O. Lecointe, J. Dento, N. Cherfaoui, V. Marion, and J. G. Morillon, "Leader-follower function for autonomous military convoys," *Proc. SPIE*, vol. 5422, pp. 326–337, Sep. 2004.
- [18] W. Liang, R. Ruhl, and J. Medanic, "Simulation of intelligent convoy with autonomous articulated commercial vehicles," SAE Tech. Paper 2003-01-3419, 2003.

- [19] R. White and M. Tomizuka, "Autonomous following lateral control of heavy vehicles using laser scanning radar," in *Proc. Amer. Control Conf.*, Jun. 2001, pp. 2333–2338.
- [20] C. Fries and H.-J. Wuensche, "Autonomous convoy driving by night: The vehicle tracking system," in *Proc. IEEE Int. Conf. Technol. Practical Robot Appl. (TePRA)*, May 2015, pp. 1–6.
- [21] M. D. Ollis, "Position estimation and vehicle control in autonomous multi-vehicle convoys," U.S. Patent 14 160 524, Oct. 16, 2014.
- [22] X. Zhao, W. Yao, N. Li, and Y. Wang, "Design of leader's path following system for multi-vehicle autonomous convoy," in *Proc. IEEE Int. Conf. Unmanned Syst. (ICUS)*, Oct. 2017, pp. 132–138.
- [23] V. Hassija, V. Chamola, G. Han, J. J. P. C. Rodrigues, and M. Guizani, "DAGIoV: A framework for vehicle to vehicle communication using directed acyclic graph and game theory," *IEEE Trans. Veh. Technol.*, vol. 69, no. 4, pp. 4182–4191, Apr. 2020.
- [24] C. Gustafson, K. Mahler, D. Bolin, and F. Tufvesson, "The COST IRA-CON geometry-based stochastic channel model for Vehicle-to-Vehicle communication in intersections," *IEEE Trans. Veh. Technol.*, vol. 69, no. 3, pp. 2365–2375, Mar. 2020.
- [25] C. Fries, T. Luettel, and H.-J. Wuensche, "Combining model- and template-based vehicle tracking for autonomous convoy driving," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Jun. 2013, pp. 1022–1027.
- [26] U. P. Mudalige and S. Zeng, "Autonomous convoying technique for vehicles," U.S. Patent 9,165,470, Oct. 20, 2015.
- [27] H. M. D. Kabir, A. Khosravi, M. A. Hosen, and S. Nahavandi, "Partial adversarial training for prediction interval," in *Proc. Int. Joint Conf. Neural Netw. (IJCNN)*, Jul. 2018, pp. 1–6.
- [28] R. Alizadehsani, M. Roshanzamir, S. Hussain, A. Khosravi, A. Koohestani, M. H. Zangooei, M. Abdar, A. Beykikhoshk, A. Shoeibi, A. Zare, M. Panahiazar, S. Nahavandi, D. Srinivasan, A. F. Atiya, and U. R. Acharya, "Handling of uncertainty in medical data using machine learning and probability theory techniques: A review of 30 years (1991–2020)," *Ann. Oper. Res.*, pp. 1–42, Mar. 2021. [Online]. Available: https://link.springer.com/article/10.1007/s10479-021-04006-2, doi: 10. 1007/s10479-021-04006-2.
- [29] D. Sadeghi, A. Shoeibi, N. Ghassemi, P. Moridian, A. Khadem, R. Alizadehsani, M. Teshnehlab, J. M. Gorriz, and S. Nahavandi, "An overview on artificial intelligence techniques for diagnosis of schizophrenia based on magnetic resonance imaging modalities: Methods, challenges, and future works," 2021, arXiv:2103.03081.
- [30] P. Petrov, "A mathematical model for control of an autonomous vehicle convoy," *Trans. Syst. Control*, vol. 3, no. 9, pp. 835–848, Sep. 2008.
- [31] P. Petrov and O. Boumbarov, "Nonlinear adaptive control of a twovehicle autonomous convoy using a look-ahead approach," in *Proc. 7th WSEAS Int. Conf. Signal Process., Robot. Autom.*, Feb. 2008, pp. 55–60.
- [32] M. Manz, T. Luettel, F. von Hundelshausen, and H.-J. Wuensche, "Monocular model-based 3D vehicle tracking for autonomous vehicles in unstructured environment," in *Proc. IEEE Int. Conf. Robot. Autom.*, May 2011, pp. 2465–2471.
- [33] H. B. Jond, J. Platosa, and Z. Sadreddini, "Autonomous vehicle convoy formation control with size/shape switching for automated highways," *Int. J. Eng.*, vol. 33, no. 11, pp. 2174–2180, 2020.
- [34] G. Lu and M. Tomizuka, "A laser scanning radar based autonomous lateral vehicle following control scheme for automated highways," in *Proc. Amer. Control Conf.*, Jun. 2003, pp. 30–35.
- [35] I. Navarro, F. Zimmermann, M. Vasic, and A. Martinoli, "Distributed graph-based control of convoys of heterogeneous vehicles using curvilinear road coordinates," in *Proc. IEEE 19th Int. Conf. Intell. Transp. Syst.* (*ITSC*), Nov. 2016, pp. 879–886.
- [36] H. Chehardoli and M. Homaienezhad, "Third-order decentralized safe consensus protocol for inter-connected heterogeneous vehicular platoons," *Int. J. Eng.*, vol. 31, no. 6, pp. 967–972, 2018.
- [37] M. Mohamed-Ahmed, A. Naamane, and N. M'sirdi, "Path tracking for the convoy of autonomous vehicles based on a non-linear predictive control," in *Proc. 12th Int. Conf. Integr. Modeling Anal. Appl. Control Autom.*, Sep. 2019, pp. 1–8.
- [38] S. Kato, S. Tsugawa, K. Tokuda, T. Matsui, and H. Fujii, "Vehicle control algorithms for cooperative driving with automated vehicles and intervehicle communications," *IEEE Trans. Intell. Transp. Syst.*, vol. 3, no. 3, pp. 155–161, Sep. 2002.
- [39] S. Gowal, R. Falconi, and A. Martinoli, "Local graph-based distributed control for safe highway platooning," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Oct. 2010, pp. 6070–6076.

- [40] X. Qian, A. de La Fortelle, and F. Moutarde, "A hierarchical model predictive control framework for on-road formation control of autonomous vehicles," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Jun. 2016, pp. 376–381.
- [41] X. Liu, S. S. Ge, and C.-H. Goh, "Vision-based leader-follower formation control of multiagents with visibility constraints," *IEEE Trans. Control Syst. Technol.*, vol. 27, no. 3, pp. 1326–1333, May 2019.
- [42] M. F. S. Rabelo, A. S. Brandao, and M. Sarcinelli-Filho, "Centralized control for an heterogeneous line formation using virtual structure approach," in *Proc. Latin Amer. Robotic Symp., Brazilian Symp. Robot.* (SBR) Workshop Robot. Educ. (WRE), Nov. 2018, pp. 135–140.
- [43] T. Balch and R. C. Arkin, "Behavior-based formation control for multirobot teams," *IEEE Trans. Robot. Autom.*, vol. 14, no. 6, pp. 926–939, Dec. 1998.
- [44] D. Xu, X. Zhang, Z. Zhu, C. Chen, and P. Yang, "Behavior-based formation control of swarm robots," *Math. Problems Eng.*, vol. 2014, pp. 1–13, Jun. 2014.
- [45] M. Khodatars, A. Shoeibi, D. Sadeghi, N. Ghaasemi, M. Jafari, P. Moridian, A. Khadem, R. Alizadehsani, A. Zare, Y. Kong, A. Khosravi, S. Nahavandi, S. Hussain, U. R. Acharya, and M. Berk, "Deep learning for neuroimaging-based diagnosis and rehabilitation of autism spectrum disorder: A review," *Comput. Biol. Med.*, vol. 139, Dec. 2021, Art. no. 104949.
- [46] A. Shoeibi, M. Khodatars, M. Jafari, P. Moridian, M. Rezaei, R. Alizadehsani, F. Khozeimeh, J. M. Gorriz, J. Heras, M. Panahiazar, S. Nahavandi, and U. R. Acharya, "Applications of deep learning techniques for automated multiple sclerosis detection using magnetic resonance imaging: A review," 2021, arXiv:2105.04881.
- [47] H. B. Jond, V. V. Nabiyev, and D. Lukáš, "Linear quadratic differential game formulation for leaderless formation control," *J. Ind. Syst. Eng.*, vol. 11, pp. 47–58, Jan. 2017.
- [48] A. Poorva, R. Gautam, and R. Kala, "Motion planning for a chain of mobile robots using A* and potential field," *Robotics*, vol. 7, no. 2, p. 20, May 2018.
- [49] Z. Huang, D. Chu, C. Wu, and Y. He, "Path planning and cooperative control for automated vehicle platoon using hybrid automata," *IEEE Trans. Intell. Transp. Syst.*, vol. 20, no. 3, pp. 959–974, Mar. 2019.
- [50] L. Gao, D. Chu, Y. Cao, L. Lu, and C. Wu, "Multi-lane convoy control for autonomous vehicles based on distributed graph and potential field," in *Proc. IEEE Intell. Transp. Syst. Conf. (ITSC)*, Oct. 2019, pp. 2463–2469.
- [51] G. S. Hornby, S. Takamura, J. Yokono, O. Hanagata, T. Yamamoto, and M. Fujita, "Evolving robust gaits with aibo," in *Proc. ICRA Millennium Conf. IEEE Int. Conf. Robot. Automat. Symposia*, vol. 3, Apr. 2000, pp. 3040–3045.
- [52] W. Kunz, "Sony corporation," in *Global Media Giants*. Evanston, IL, USA: Routledge, 2016, pp. 239–253.
- [53] S. Lam and J. Katupitiya, "Cooperative intersection negotiation for multiple autonomous platoons," *IFAC Proc. Volumes*, vol. 46, no. 10, pp. 48–53, Jun. 2013.
- [54] Auto. Mobility Appliqué System (AMAS). Accessed: Sep. 2, 2020. [Online]. Available: https://www.lockheedmartin.com/en-us/products/ autonomous-mobility-appli%que-system-amas.html
- [55] G. Li, Y. Yang, T. Zhang, X. Qu, D. Cao, B. Cheng, and K. Li, "Risk assessment based collision avoidance decision-making for autonomous vehicles in multi-scenarios," *Transp. Res. C, Emerg. Technol.*, vol. 122, Jan. 2021, Art. no. 102820.
- [56] M. Moghadam and G. H. Elkaim, "An autonomous driving framework for long-term decision-making and short-term trajectory planning on frenet space," in *Proc. IEEE 17th Int. Conf. Autom. Sci. Eng. (CASE)*, Aug. 2021, pp. 1745–1750.
- [57] F. Camara, N. Bellotto, S. Cosar, F. Weber, D. Nathanael, M. Althoff, J. Wu, J. Ruenz, A. Dietrich, G. Markkula, A. Schieben, F. Tango, N. Merat, and C. Fox, "Pedestrian models for autonomous driving—Part II: high-level models of human behavior," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 9, pp. 5453–5472, Sep. 2021.
- [58] H. R. M. Pelikan, "Why autonomous driving is so hard: The social dimension of traffic," in *Proc. Companion ACM/IEEE Int. Conf. Hum.-Robot Interact.*, Mar. 2021, pp. 81–85.
- [59] M. Stryszowski, S. Longo, E. Velenis, and G. Forostovsky, "A framework for self-enforced interaction between connected vehicles: Intersection negotiation," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 11, pp. 6716–6725, Nov. 2021.

- [60] J. Hook, S. El-Sedky, V. D. Silva, and A. Kondoz, "Learning data-driven decision-making policies in multi-agent environments for autonomous systems," *Cognit. Syst. Res.*, vol. 65, pp. 40–49, Jan. 2021.
- [61] L. Liebenwein, W. Schwarting, C.-I. Vasile, J. DeCastro, J. Alonso-Mora, S. Karaman, and D. Rus, "Compositional and contract-based verification for autonomous driving on road networks," in *Robotics Research*. Cham, Switzerland: Springer, 2020, pp. 163–181.
- [62] P. Cai, S. Wang, Y. Sun, and M. Liu, "Probabilistic end-to-end vehicle navigation in complex dynamic environments with multimodal sensor fusion," *IEEE Robot. Autom. Lett.*, vol. 5, no. 3, pp. 4218–4224, Jul. 2020.
- [63] P. Hang, C. Lv, Y. Xing, C. Huang, and Z. Hu, "Human-like decision making for autonomous driving: A noncooperative game theoretic approach," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 4, pp. 2076–2087, Apr. 2021.
- [64] J. Shen, J. Y. Won, Z. Chen, and Q. A. Chen, "Drift with devil: Security of multi-sensor fusion based localization in high-level autonomous driving under GPS spoofing," in Proc. 29th USENIX Secur. Symp. (USENIX Secur.), 2020, pp. 931–948.
- [65] J.-W. Hu, B.-Y. Zheng, C. Wang, C.-H. Zhao, X.-L. Hou, Q. Pan, and Z. Xu, "A survey on multi-sensor fusion based obstacle detection for intelligent ground vehicles in off-road environments," *J. Frontiers Inf. Technol. Electron. Eng.*, vol. 21, no. 5, pp. 675–692, May 2020.
- [66] Y. Choi, N. Kim, S. Hwang, K. Park, J. S. Yoon, K. An, and I. S. Kweon, "KAIST multi-spectral day/night data set for autonomous and assisted driving," *IEEE Trans. Intell. Transp. Syst.*, vol. 19, no. 3, pp. 934–948, Mar. 2018.
- [67] B. Theisen, A. Lester, L. Perecko, and R. Folden, "Autonomous mobility appliqué system joint capability technology demonstration," Army Tank Automot. Res. Develop. Eng. Center Warren, Warren, MI, Tech. Rep. ADA579685, 2013.
- [68] A. Koohestani, M. Abdar, S. Hussain, A. Khosravi, D. Nahavandi, S. Nahavandi, and R. Alizadehsani, "Analysis of driver performance using hybrid of weighted ensemble learning technique and evolutionary algorithms," *Arabian J. Sci. Eng.*, vol. 46, no. 4, pp. 3567–3580, Apr. 2021.
- [69] R. Ardhi, M. R. Febsya, A. Widyotriatmo, and Y. Y. Nazaruddin, "Backward motion path following control of autonomous truck-trailer: Lyapunov stability approach," in *Proc. IEEE Conf. Control Technol. Appl.* (CCTA), Aug. 2019, pp. 900–905.
- [70] S. Ucar, S. C. Ergen, and O. Ozkasap, "IEEE 802.11p and visible light hybrid communication based secure autonomous platoon," *IEEE Trans. Veh. Technol.*, vol. 67, no. 9, pp. 8667–8681, Sep. 2018.
- [71] Practical Autonomous Trucking for Today. Accessed: Nov. 2, 2020. [Online]. Available: https://locomation.ai/
- [72] L. Zhang, F. Chen, X. Ma, and X. Pan, "Fuel economy in truck platooning: A literature overview and directions for future research," *J. Adv. Transp.*, vol. 2020, pp. 1–10, Jan. 2020.
- [73] We've Got a Mighty Convoy: How Driverless Trucks Will One Day Rule Road. Accessed: Sep. 2, 2020. [Online]. Available: https://www.orangebusiness.com/en/blogs/connecting-technology/m2m/we-v%e-got-amighty-convoy-how-driverless-trucks-will-one-day-rule-the-road
- [74] Truck platooning on European Roads. Accessed: Sep. 2, 2020. [Online]. Available: https://www.volvogroup.com/en-en/news/2018/feb/truckplatooning-on-euro%pean-roads.html
- [75] Waymo's Self-Driving Trucks Will Start Delivering Freight Atlanta. Accessed: Sep. 2, 2020. [Online]. Available: https://www.theverge. com/2018/3/9/17100518/waymo-self-driving-truck-goo%gle-atlanta
- [76] National Robotics Engineering Center. Accessed: Sep. 2, 2020. [Online]. Available: https://www.nrec.ri.cmu.edu/nrec/solutions/defense/cargougv.html
- [77] Global Autonomous Military Vehicle Market 2018–2022 Key Vendors are BAE Systems, Elbit Systems, Lockheed Martin, Northrop Grumman, Polaris Industries & RUAG Group ResearchAndMarkets.Com. Accessed: Sep. 2, 2020. [Online]. Available: https://www.businesswire. com/news/home/20180830005654/en/Global-Autonom%ous-Military-Vehicle-Market-2018-2022–
- [78] J. C. Elsey, "Data compression for space missions," in *Proc. Space Congr.*, Apr. 1968, pp. 1–15.
- [79] H. Wilhelm, "The development of German army airplanes during the war," NASA, Washington, DC, USA, Tech. Rep. NACA TN 56, 1921.
- [80] Army Wants 70 Self-Driving Supply Trucks By 2020. Accessed: Sep. 2, 2020. [Online]. Available: https://breakingdefense.com/2018/08/ army-wants-70-self-driving-supply-t%rucks-by-2020/

- [81] H. M. D. Kabir, A. Khosravi, S. K. Mondal, M. Rahman, S. Nahavandi, and R. Buyya, "Uncertainty-aware decisions in cloud computing: Foundations and future directions," *ACM Comput. Surv.*, vol. 54, no. 4, pp. 1–30, May 2022.
- [82] F. Nasirzadeh, H. M. D. Kabir, M. Akbari, A. Khosravi, S. Nahavandi, and D. G. Carmichael, "ANN-based prediction intervals to forecast labour productivity," *Eng., Construct. Architectural Manage.*, vol. 27, no. 9, pp. 2335–2351, May 2020.
- [83] O. E. M. J. Musa, S. Sudin, S. Garba, S. A. Sala, and A. A. Oremeyi, "Internet of vehicle for two-vehicle look-ahead convoy system using state feedback control," in *Machine Learning and Data Mining for Emerging Trend in Cyber Dynamics: Theories and Applications.* Cham, Switzerland: Springer, 2021, p. 241.
- [84] S. K. Mondal, R. Pan, H. D. Kabir, T. Tian, and H.-N. Dai, "Kubernetes in it administration and serverless computing: An empirical study and research challenges," J. Supercomput., vol. 78, pp. 2937–2987, Jul. 2021.
- [85] A. E. Opcin, A. H. Buss, T. W. Lucas, and P. J. Sanchez, "Modeling anti-air warfare with discrete event simulation and analyzing naval convoy operations," in *Proc. Winter Simulation Conf. (WSC)*, Dec. 2017, pp. 4048–4057.
- [86] O. Vinyals et al., "Grandmaster level in starcraft II using multi-agent reinforcement learning," Nature, vol. 575, no. 7782, pp. 350–354, 2019.
- [87] F. Albardi, H. M. D. Kabir, M. M. I. Bhuiyan, P. M. Kebria, A. Khosravi, and S. Nahavandi, "A comprehensive study on torchvision pre-trained models for fine-grained inter-species classification," in *Proc. IEEE Int. Conf. Syst., Man, Cybern. (SMC)*, Oct. 2021, pp. 2767–2774.
- [88] H. M. D. Kabir, A. Khosravi, S. Nahavandi, and A. Kavousi-Fard, "Partial adversarial training for neural network-based uncertainty quantification," *IEEE Trans. Emerg. Topics Comput. Intell.*, vol. 5, no. 4, pp. 595–606, Aug. 2021.
- [89] Y. Rizk, M. Awad, and E. W. Tunstel, "Cooperative heterogeneous multirobot systems: A survey," ACM Comput. Surv., vol. 52, no. 2, pp. 1–31, 2019.
- [90] DeepMind—The Role of Multi-Agent Learning in Artificial Intelligence Research. Accessed: Sep. 2, 2020. [Online]. Available: https://www.youtube.com/watch?v=yE62Zwhmzi8&t=2279s
- [91] M. Kamali, L. A. Dennis, O. McAree, M. Fisher, and S. M. Veres, "Formal verification of autonomous vehicle platooning," *Sci. Comput. Program.*, vol. 148, pp. 88–106, Nov. 2017.
- [92] M. Segata, F. Dressler, R. Lo Cigno, and M. Gerla, "A simulation tool for automated platooning in mixed highway scenarios," in *Proc. 18th Annu. Int. Conf. Mobile Comput. Netw. (Mobicom)*, 2012, pp. 389–392.
- [93] F. Gechter, J.-M. Contet, S. Galland, O. Lamotte, and A. Koukam, "Virtual intelligent vehicle urban simulator: Application to vehicle platoon evaluation," *Simul. Model. Pract. Theory*, vol. 24, pp. 103–114, May 2012.
- [94] Z. Chen, J. Zhang, and D. Tao, "Progressive LiDAR adaptation for road detection," *IEEE/CAA J. Automatica Sinica*, vol. 6, no. 3, pp. 693–702, May 2019.
- [95] Visual Perception for Self-Driving Cars, Lesson 3: Semantic Segmentation for Road Scene Understanding. Accessed: Sep. 2, 2020. [Online]. Available: https://www.coursera.org/lecture/visual-perceptionself-driving-cars/le%sson-3-semantic-segmentation-for-road-sceneunderstanding-RDi8V
- [96] D. Riehle, D. Reiser, and H. W. Griepentrog, "Robust index-based semantic plant/background segmentation for RGB-images," *Comput. Electron. Agricult.*, vol. 169, Feb. 2020, Art. no. 105201.
- [97] A. Roy and S. Todorovic, "A multi-scale CNN for affordance segmentation in RGB images," in *Proc. Eur. Conf. Comput. Vis.* Cham, Switzerland: Springer, 2016, pp. 186–201.
- [98] A. Milioto, I. Vizzo, J. Behley, and C. Stachniss, "RangeNet++: Fast and accurate LiDAR semantic segmentation," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst. (IROS)*, Nov. 2019, pp. 4213–4220.
- [99] E. E. Aksoy, S. Baci, and S. Cavdar, "SalsaNet: Fast road and vehicle segmentation in LiDAR point clouds for autonomous driving," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Oct. 2020, pp. 926–932.
- [100] I. Orr, M. Cohen, and Z. Zalevsky, "High-resolution radar road segmentation using weakly supervised learning," *Nature Mach. Intell.*, vol. 3, no. 3, pp. 239–246, Mar. 2021.
- [101] L. Sless, B. E. Shlomo, G. Cohen, and S. Oron, "Road scene understanding by occupancy grid learning from sparse radar clusters using semantic segmentation," in *Proc. IEEE/CVF Int. Conf. Comput. Vis. Workshop* (ICCVW), Oct. 2019, pp. 867–875.
- [102] The KITTI Vision Benchmark Suite. Accessed: Sep. 2, 2021. [Online]. Available: www.cvlibs.net/datasets/kitti/eval_road.php

- [103] H. Wang, R. Fan, P. Cai, and M. Liu, "SNE-RoadSeg+: Rethinking depthnormal translation and deep supervision for freespace detection," 2021, arXiv:2107.14599.
- [104] H. Wang, R. Fan, Y. Sun, and M. Liu, "Dynamic fusion module evolves drivable area and road anomaly detection: A benchmark and algorithms," *IEEE Trans. Cybern.*, early access, Mar. 24, 2021, doi: 10.1109/TCYB.2021.3064089.
- [105] R. Fan, H. Wang, P. Cai, and M. Liu, "SNE-RoadSeg: Incorporating surface normal information into semantic segmentation for accurate freespace detection," in *Proc. Eur. Conf. Comput. Vis.* Cham, Switzerland: Springer, 2020, pp. 340–356.
- [106] F. Castanedo, "A review of data fusion techniques," Sci. World J., vol. 2013, pp. 1–19, Sep. 2013.
- [107] M. Bojarski, P. Yeres, A. Choromanska, K. Choromanski, B. Firner, L. Jackel, and U. Müller, "Explaining how a deep neural network trained with end-to-end learning steers a car," 2017, arXiv:1704.07911.
- [108] H. D. Kabir, A. Khosravi, M. A. Hosen, and S. Nahavandi, "Neural network-based uncertainty quantification: A survey of methodologies and applications," *IEEE Access*, vol. 6, pp. 36218–36234, 2018.
- [109] K. Posch, J. Steinbrener, and J. Pilz, "Variational inference to measure model uncertainty in deep neural networks," 2019, arXiv:1902.10189.
- [110] B. V. Dasarathy, "Sensor fusion potential exploitation-innovative architectures and illustrative applications," *Proc. IEEE*, vol. 85, no. 1, pp. 24–38, Jan. 1997.
- [111] W. D. Ross, A. M. Waxman, W. W. Streilein, M. Aguiiar, J. Verly, F. Liu, M. I. Braun, P. Harmon, and S. Rak, "Multi-sensor 3D image fusion and interactive search," in *Proc. 3rd Int. Conf. Inf. Fusion*, Jul. 2000, pp. 1–8.
- [112] F. Franceschini, M. Galetto, D. Maisano, and L. Mastrogiacomo, "Combining multiple large volume metrology systems: Competitive versus cooperative data fusion," *Precis. Eng.*, vol. 43, pp. 514–524, Jan. 2016.
- [113] Y. Dobrev, S. Flores, and M. Vossiek, "Multi-modal sensor fusion for indoor mobile robot pose estimation," in *Proc. IEEE/ION Position, Location Navigat. Symp. (PLANS)*, Apr. 2016, pp. 553–556.
- [114] F. Khozeimeh, D. Sharifrazi, N. H. Izadi, J. H. Joloudari, A. Shoeibi, R. Alizadehsani, J. M. Gorriz, S. Hussain, Z. A. Sani, H. Moosaei, A. Khosravi, S. Nahavandi, and S. M. S. Islam, "Combining a convolutional neural network with autoencoders to predict the survival chance of COVID-19 patients," *Sci. Rep.*, vol. 11, no. 1, pp. 1–18, Dec. 2021.
- [115] A. Shoeibi, N. Ghassemi, M. Khodatars, M. Jafari, P. Moridian, R. Alizadehsani, A. Khadem, Y. Kong, A. Zare, J. M. Gorriz, J. Ramírez, M. Panahiazar, A. Khosravi, and S. Nahavandi, "Applications of epileptic seizures detection in neuroimaging modalities using deep learning techniques: Methods, challenges, and future works," 2021, arXiv:2105.14278.
- [116] J. Kocic, N. Jovicic, and V. Drndarevic, "Sensors and sensor fusion in autonomous vehicles," in *Proc. 26th Telecommun. Forum (TELFOR)*, Nov. 2018, pp. 420–425.
- [117] R. P. D. Vivacqua, M. Bertozzi, P. Cerri, F. N. Martins, and R. F. Vassallo, "Self-localization based on visual lane marking maps: An accurate lowcost approach for autonomous driving," *IEEE Trans. Intell. Transp. Syst.*, vol. 19, no. 2, pp. 582–597, Feb. 2018.
- [118] H. M. D. Kabir, A. Khosravi, D. Nahavandi, and S. Nahavandi, "Uncertainty quantification neural network from similarity and sensitivity," in *Proc. Int. Joint Conf. Neural Netw. (IJCNN)*, Jul. 2020, pp. 1–8.
- [119] M. R. C. Qazani, H. Asadi, T. Bellmann, S. Perdrammehr, S. Mohamed, and S. Nahavandi, "A new fuzzy logic based adaptive motion cueing algorithm using parallel simulation-based motion platform," in *Proc. IEEE Int. Conf. Fuzzy Syst. (FUZZ-IEEE)*, Jul. 2020, pp. 1–8.
- [120] H. M D. Kabir, A. Khosravi, A. Kavousi-Fard, S. Nahavandi, and D. Srinivasan, "Optimal uncertainty-guided neural network training," 2019, arXiv:1912.12761.
- [121] F. Verdoja, J. Lundell, and V. Kyrki, "Deep network uncertainty maps for indoor navigation," in *Proc. IEEE-RAS 19th Int. Conf. Humanoid Robots* (*Humanoids*), Oct. 2019, pp. 112–119.
- [122] P. Stepan, M. Kulich, and L. Preucil, "Robust data fusion with occupancy grid," *IEEE Trans. Syst., Man, C, Appl. Rev.*, vol. 35, no. 1, pp. 106–115, Feb. 2005.
- [123] M. Ribo and A. Pinz, "A comparison of three uncertainty calculi for building sonar-based occupancy grids," *Robot. Auto. Syst.*, vol. 35, nos. 3–4, pp. 201–209, 2001.
- [124] Military Sensor Fusion Market Set to Grow to \$756mn by 2029. Accessed: Sep. 2, 2020. [Online]. Available: https://www.asdnews.com/news/ defense/2018/12/03/military-sensor-fusion-%market-set-grow-756mn-2029

- [125] M. Panicker, T. Mitha, K. Oak, A. M. Deshpande, and C. Ganguly, "Multisensor data fusion for an autonomous ground vehicle," in *Proc. Conf. Adv. Signal Process. (CASP)*, Jun. 2016, pp. 507–512.
- [126] Lockheed Martin To Develop Next-Gen Sensor Fusion Testbed to Enhance Helicopter Survivability. Accessed: Sep. 2, 2020. [Online]. Available: https://news.lockheedmartin.com/2017-10-17-Lockheed-Martin-to-Develop-N%ext-Gen-Sensor-Fusion-Testbed-to-Enhance-Helicopter-Survivability
- [127] M. Song, F. Chen, and X. Ma, "Organization of autonomous truck platoon considering energy saving and pavement fatigue," *Transp. Res. D*, *Transp. Environ.*, vol. 90, Jan. 2021, Art. no. 102667.
- [128] D. Liu, B. Eksioglu, M. J. Schmid, N. Huynh, and G. Comert, "Optimizing energy savings for a fleet of commercial autonomous trucks," *IEEE Trans. Intell. Transp. Syst.*, early access, Apr. 16, 2021, doi: 10.1109/TITS.2021.3071442.
- [129] Z. Ying, M. Ma, Z. Zhao, X. Liu, and J. Ma, "A reputation-based leader election scheme for opportunistic autonomous vehicle platoon," *IEEE Trans. Veh. Technol.*, early access, Aug. 20, 2021, doi: 10.1109/TVT.2021.3106297.
- [130] O. El Ganaoui-Mourlan, S. Camp, T. Hannagan, V. Arora, M. D. Neuville, and V. A. Kousournas, "Path planning for autonomous platoon formation," *Sustainability*, vol. 13, no. 9, p. 4668, Apr. 2021.
- [131] E. F. Z. Santana, G. Covas, F. Duarte, P. Santi, C. Ratti, and F. Kon, "Transitioning to a driverless city: Evaluating a hybrid system for autonomous and non-autonomous vehicles," *Simul. Model. Pract. Theory*, vol. 107, Feb. 2021, Art. no. 102210.
- [132] H. Manivasakan, R. Kalra, S. O'Hern, Y. Fang, Y. Xi, and N. Zheng, "Infrastructure requirement for autonomous vehicle integration for future urban and suburban roads–current practice and a case study of melbourne, australia," *Transp. Res. A, Policy Pract.*, vol. 152, pp. 36–53, 2021.
- [133] L. Zhu, Y. Tang, and D. Yang, "Cellular automata-based modeling and simulation of the mixed traffic flow of vehicle platoon and normal vehicles," *Phys. A, Stat. Mech. Appl.*, vol. 584, Dec. 2021, Art. no. 126368.
- [134] H. M. D. Kabir, S. B. Alam, M. I. Azam, M. A. Hussain, A. B. M. R. Sazzad, M. N. Sakib, and M. A. Matin, "Non-linear downsampling and signal reconstruction, without folding," in *Proc. 4th UKSim Eur. Symp. Comput. Modeling Simulation*, Nov. 2010, pp. 142–146.
- [135] L.-H. Li, J. Gan, and W.-Q. Li, "A separation strategy for connected and automated vehicles: Utilizing traffic light information for reducing idling at red lights and improving fuel economy," *J. Adv. Transp.*, vol. 2018, pp. 1–10, Jul. 2018.
- [136] M. Liu, M. Wang, and S. Hoogendoorn, "Optimal platoon trajectory planning approach at arterials," *Transp. Res. Rec.*, vol. 2673, no. 9, pp. 214–226, 2019.
- [137] M. Siam, S. Elkerdawy, M. Jagersand, and S. Yogamani, "Deep semantic segmentation for automated driving: Taxonomy, roadmap and challenges," in *Proc. IEEE 20th Int. Conf. Intell. Transp. Syst. (ITSC)*, Oct. 2017, pp. 1–8.
- [138] E. Yurtsever, J. Lambert, A. Carballo, and K. Takeda, "A survey of autonomous driving: Common practices and emerging technologies," *IEEE Access*, vol. 8, pp. 58443–58469, 2020.
- [139] J. Rissman *et al.*, "Technologies and policies to decarbonize global industry: Review and assessment of mitigation drivers through 2070," *Appl. Energy*, vol. 266, May 2020, Art. no. 114848.
- [140] P. Kohli and A. Chadha, "Enabling pedestrian safety using computer vision techniques: A case study of the 2018 Uber Inc. Self-driving car crash," in *Proc. Future Inf. Commun. Conf.* Cham, Switzerland: Springer, 2019, pp. 261–279.
- [141] V. C. M. Vishnu, M. Rajalakshmi, and R. Nedunchezhian, "Intelligent traffic video surveillance and accident detection system with dynamic traffic signal control," *Cluster Comput.*, vol. 21, no. 1, pp. 135–147, 2018.
- [142] M. Kaushal, B. S. Khehra, and A. Sharma, "Soft computing based object detection and tracking approaches: State-of-the-art survey," *Appl. Soft Comput.*, vol. 70, pp. 423–464, Sep. 2018.
- [143] A. Garcia-Garcia, S. Orts-Escolano, S. Oprea, V. Villena-Martinez, P. Martinez-Gonzalez, and J. Garcia-Rodriguez, "A survey on deep learning techniques for image and video semantic segmentation," *Appl. Soft Comput.*, vol. 70, pp. 41–65, Sep. 2018.
- [144] U. M. Gidado, H. Chiroma, N. Aljojo, S. Abubakar, S. I. Popoola, and M. A. Al-Garadi, "A survey on deep learning for steering angle prediction in autonomous vehicles," *IEEE Access*, vol. 8, pp. 163797–163817, 2020.
- [145] F. Codevilla, E. Santana, A. Lopez, and A. Gaidon, "Exploring the limitations of behavior cloning for autonomous driving," in *Proc. IEEE/CVF Int. Conf. Comput. Vis. (ICCV)*, Oct. 2019, pp. 9329–9338.

- [146] A. Artunedo, J. Godoy, and J. Villagra, "Smooth path planning for urban autonomous driving using OpenStreetMaps," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Jun. 2017, pp. 837–842.
- [147] A. Dosovitskiy, G. Ros, F. Codevilla, A. Lopez, and V. Koltun, "Carla: An open urban driving simulator," in *Proc. Conf. Robot Learn.*, 2017, pp. 1–16.
- [148] S. Hossain and D.-J. Lee, "Autonomous-driving vehicle learning environments using unity real-time engine and end-to-end CNN approach," *J. Korea Robot. Soc.*, vol. 14, no. 2, pp. 122–130, May 2019.
 [149] N. Cruz, L. G. Marin, and D. Saez, "Prediction intervals with LSTM
- [149] N. Cruz, L. G. Marin, and D. Saez, "Prediction intervals with LSTM networks trained by joint supervision," in *Proc. Int. Joint Conf. Neural Netw. (IJCNN)*, Jul. 2019, pp. 1–8.
- [150] P. Morala, J. A. Cifuentes, R. E. Lillo, and I. Ucar, "Towards a mathematical framework to inform neural network modelling via polynomial regression," *Neural Netw.*, vol. 142, pp. 57–72, Oct. 2021.
- [151] E. Javanmardi, S. Liu, and N. Xie, "Exploring the philosophical foundations of grey systems theory: Subjective processes, information extraction and knowledge formation," *Found. Sci.*, vol. 26, no. 2, pp. 371–404, Jun. 2021.
- [152] M. Hossain, S. Mekhilef, M. Danesh, L. Olatomiwa, and S. Shamshirband, "Application of extreme learning machine for short term output power forecasting of three grid-connected PV systems," *J. Cleaner Prod.*, vol. 167, pp. 395–405, Nov. 2017.
- [153] M. S. Kamal, M. G. Sarowar, N. Dey, A. S. Ashour, S. H. Ripon, B. K. Panigrahi, and J. M. R. S. Tavares, "Self-organizing mapping based swarm intelligence for secondary and tertiary proteins classification," *Int. J. Mach. Learn. Cybern.*, vol. 10, no. 2, pp. 229–252, Feb. 2019.
- [154] C. Serpell, I. Araya, C. Valle, and H. Allende, "Probabilistic forecasting using Monte Carlo dropout neural networks," in *Iberoamerican Congress on Pattern Recognition*. Cham, Switzerland: Springer, 2019, pp. 387–397.
- [155] H. M D. Kabir, M. Abdar, S. M. J. Jalali, A. Khosravi, A. F Atiya, S. Nahavandi, and D. Srinivasan, "SpinalNet: Deep neural network with gradual input," 2020, arXiv:2007.03347.
- [156] S. Liu, A. Borovykh, L. A. Grzelak, and C. W. Oosterlee, "A neural network-based framework for financial model calibration," *J. Math. Ind.*, vol. 9, no. 1, pp. 1–28, Dec. 2019.
- [157] V. V. Kindratenko and W. R. Sherman, "Neural network-based calibration of electromagnetic tracking systems," *Virtual Reality*, vol. 9, no. 1, pp. 70–78, Dec. 2005.
- [158] P. M. Kebria, A. Khosravi, S. M. Salaken, I. Hossain, H. D. Kabir, A. Koohestani, R. Alizadehsani, and S. Nahavandi, "Deep imitation learning: The impact of depth on policy performance," in *Proc. Int. Conf. Neural Inf. Process.* Cham, Switzerland: Springer, 2018, pp. 172–181.
- [159] P. M. Kebria, A. Khosravi, I. Hossain, N. Mohajer, H. D. Kabir, S. M. J. Jalali, D. Nahavandi, S. M. Salaken, S. Nahavandi, A. Lagrandcourt, and N. Bhasin, "Autonomous navigation via deep imitation and transfer learning: A comparative study," in *Proc. IEEE Int. Conf. Syst.*, *Man, Cybern. (SMC)*, Oct. 2020, pp. 2907–2912.
- [160] Y. Zhu, C. Li, and Y. Zhang, "Online camera-LiDAR calibration with sensor semantic information," in *Proc. IEEE Int. Conf. Robot. Autom.* (*ICRA*), May 2020, pp. 4970–4976.
- [161] V. Cimini, E. Polino, M. Valeri, I. Gianani, N. Spagnolo, G. Corrielli, A. Crespi, R. Osellame, M. Barbieri, and F. Sciarrino, "Calibration of multiparameter sensors via machine learning at the single-photon level," *Phys. Rev. A, Gen. Phys.*, vol. 15, no. 4, Apr. 2021, Art. no. 044003.



SAEID NAHAVANDI (Fellow, IEEE) received the Ph.D. degree from Durham University, U.K., in 1991.

He is currently an Alfred Deakin Professor, a Pro Vice-Chancellor, the Chair of engineering, and the Founding and Current Director of the Institute for Intelligent Systems Research and Innovation, Deakin University. He has published over 1000 scientific papers in various international journals and conferences. His research interests

include modeling of complex systems, robotics, and haptics.

Prof. Nahavandi is a fellow of Engineers Australia (Feast) and the Institution of Engineering and Technology (FIET). He is a fellow of the Australian Academy of Technology and Engineering (ATSE). He is the Senior Associate Editor of IEEE Systems JOURNAL, an Associate Editor of IEEE TRANSACTIONS ON CYBERNETICS, and an IEEE PRESS Editorial Board Member.



SHADY MOHAMED received the Ph.D. degree from Deakin University, Australia, in 2009. He is currently an Associate Professor in signals and systems at the Institute for Intelligent Systems Research and Innovation (IISRI), Deakin University. His interests include signal processing, human physiological data analysis, autonomous systems, and motion simulation.



IBRAHIM HOSSAIN received the Ph.D. degree from Deakin University, in 2018. He is currently a Research Fellow in AI, robotics, and computational modeling at the Institute for Intelligent Systems Research and Innovation, Deakin University. He has academic and industry scale research experiences in AI, robotics, data science, and software development. His research interests include robotics, deep learning, machine learning modeling, sensor fusion, self-driving car, brain-computer interface, and big data.



DARIUS NAHAVANDI (Member, IEEE) received the Ph.D. degree from Deakin University, in 2018. He is currently a Research Fellow in human factors modeling and simulation at the Institute for Intelligent Systems Research and Innovation (IISRI), Deakin University. His interests include human physiological data analysis, autonomous systems, and assistive exoskeleton systems.



SYED MOSHFEQ SALAKEN received the Ph.D. degree in artificial intelligence from Deakin University, in 2018. He is currently a Data Science Practitioner focusing on deriving business value using machine learning. His research interests include transfer learning, explaining predictions made by AI/ML systems, and machine learning in general.



MOHAMMAD ROKONUZZAMAN received the

B.Sc. degree in electrical and electronic engineering from the Islamic University of Technology (IUT), Bangladesh, and the M.Sc. degree in space science and technology with the specialization in space robotics and automation from Aalto University, Finland, in 2015. He is currently pursuing the Ph.D. degree with the Institute for Intelligent Systems Research and Innovation (IISRI), Deakin University, Australia. His research inter-

ests include control of the autonomous vehicle, human effects in autonomous driving, and learning-based control of autonomous and semiautonomous systems.

IEEE Access



RACHAEL AYOUB graduated from the Royal Military College, Duntroon, in 1999. She received the master's degree in business and the master's degree in capability management from UNSW, Canberra, in 2019. She is currently the Developing Technology Staff Officer within the Robotic and Autonomous Systems Implementation and Coordination Office (RICO), Army's Future Land Warfare. She is a Career Army Officer and was commissioned into the Royal Australian Electrical

and Mechanical Engineers. She has worked in a range of command and logistic appointments, including a Troop Commander in East Timor with the UN sustainment forces, a Platoon Commander at the Army Recruit Training Centre, Kapooka, a Logistics Operations Officer at the Third and 11th Combat Service Support Battalion, the Team Leader of the Queensland Maintenance Advisory Team, and a Planner at the Deployable Joint Force Headquarters, where she deployed to Nepal and Fiji in support of humanitarian assistance missions.



ROBIN SMITH is currently the Inaugural Director of the Army's Robotic and Autonomous Systems Implementation and Coordination Office (RICO) within the Future Land Warfare Branch of the Australian Army HQ. RICO leads concept development around emerging and disruptive technology, including autonomy and AI, quantum technologies, and alternative power and energy. He has been overseeing army's exploration of autonomous systems, since 2017, after he has

authored the *Robotic and Autonomous Systems Strategy* (Australian Army). He is a Logistician by background and served for over 30 years in the British Army before transferring to the Australian Army. He has commanded at all levels up to and including regimental level and been deployed operationally a number of times, including Afghanistan multiple times, Sierra Leone, and supported other deployments for which he was awarded the OBE, in 2013. He has a background in electronic engineering and has fulfilled a number of roles in the joint logistics domain at the tactical and operational levels.

. . .